EXPLORING LECTURERS' EXPERIENCES OF TEACHING LEVEL 2 LIFE

ORIENTATION (COMPUTER SKILLS) IN NATIONAL CERTIFICATE

VOCATIONAL AT A TECHNICAL VOCATIONAL EDUCATIONAL AND

TRAINING (TVET) COLLEGE IN A TOWNSHIP

by

A.U. PAUL

Submitted in partial fulfilment of the academic requirements for the degree of Master of Education School of Education, College of Humanities University of KwaZulu-Natal

ABSTRACT

The focus and purpose of the study was to explore Life Orientation lecturers' experiences in teaching Computer Skills at a township Technical Vocational Education and Training (TVET) College. The study explored the experiences of Life Orientation lecturers that taught Level 2 Computer Skills. The research employed the interpretivist paradigm. The qualitative approach was used which allowed for a greater knowledge and understanding of the lecturers' experiences. Data was generated using semi-structured interviews with the four purposively selected Life Orientation lecturers who taught Level 2 Computer Skills. The study concluded that lecturers were struggling with Computer Skills from the outset of National Curriculum Vocational (NCV) implementation and numerous technology challenges persisted thereafter. Some lecturers believed that they had to migrate to a different world without the requisite support. The requisite support would allow them to learn and internalise aspects of Computer Skills which made them not confident in a higher education context. This study consequently recommended the training of lecturers by the Department of Higher Education and Training before they teach Computer Skills within the curriculum. The College should provide an on-going support structure as a higher education institution to ensure that the lecturers receive sufficient assistance to progress to teaching Levels 3 and 4.

DECLARATION

I, Amanda Paul, declare that the research reported in this dissertation for Master of Education, titled: Exploring lecturers' experiences of teaching Level 2 Life **Orientation (computer skills) in National Certificate Vocational at a Technical** Vocational Educational and Training (TVET) college in a township, is my own work; and all sources used have been acknowledged in-text and in the reference list accordingly.

Signature: Freu]

Date: 26 April 2019

Amanda Paul Student number: 875874211

As the candidate's Supervisor I agree/do not agree to the submission of this dissertation.

Dr Nomkhosi Nzimande

Signature:

Date: 26 April 2019

TABLE OF CONTENTS

ABST	RACT	ii
DECLARATION		
LIST	OF TABLES	vi
LIST	OF ACRONYMS	vii
ACKNOWLEDGEMENT		
DEDI	CATION	X
CHAF	TER 1 INTRODUCTION AND BACKGROUND TO THE STUDY	
1.1	Introduction	1
1.2	Background to the study	1-2
1.3	Purpose of the study	2-3
1.4	Research problem	3-4
1.5	Research question	4
1.6	Organisation of chapters	4-5
1.7	Conclusion	6
CHAF	TER 2 REVIEW OF RELATED LITERATURE AND THEORETICAL	
0.1	FRAMEWORK	7
2.1	Introduction	7
2.2	International teaching of computer skills	7.0
2.2.1	National teaching of computer skills	/-9 0.11
2.2.2	TVET education (learner controd)	9-11
2.5	TVET education (life long learning)	11-12
2.4	TVET education (me-long learning)	12-15
2.5	Theoretical framework	13-14
2.0 2.6.1	TPACK framework (technological nedagogical and content knowledge)	14 14_16
2.0.1	Technological knowledge	16_17
2.0.1.1		10-1/

2.6.1.2	2 Pedagogical knowledge	17-19
2.6.1.3	3 Content knowledge	19
2.7	Conclusion	19-20

CHAPTER 3 RESEARCH METHOD

3.1	Introduction	21
3.2	Research design	21
3.2.1	Interpretive paradigm	21-22
3.2.2	Qualitative approach	22-23
3.3	Research methodology	23
3.3.1	Case study	23-24
3.4	Data generation	24
3.4.1	Semi-structured interviews	24-27
3.5	Selection of participants	27
3.5.1	Sampling	27
3.5.2	Participants' profile	28
3.6	Data analysis process	28-29
3.7	Ethical considerations	29-30
3.8	Trustworthiness and credibility	30-31
3.9	Transferability and dependability	31-32
3.10	Limitations of the study	32
3.11	Conclusion	32-33

CHAPTER 4 DATA PRESENTATION AND ANALYSIS

4.1	Introduction	34
4.2	Thematic analysis of data	34-35
4.2.1	Theme one: Familiarity and non-familiarity with teaching approaches	35-37
4.2.2	Theme two: Availability and unavailability of adequate resources	37-40
4.2.3	Theme three: Lecturer training and professional development supported	
	by DHET	40-44
4.2.4	Theme four: Experiences of teaching	44-47
4.2.5	Theme five: Time allocation for teaching and learning	47-49
4.2.6	Theme six: Lecturer communication and collaboration	49-51
4.2.7	Theme seven: Computer skills as a vocational subject	51-53

4.2.8	Theme eight: Subject specialisation	
4.2.9	9 Theme nine: Status of computer skills in comparison to gateway	
	subjects in NCV	55-57
4.3	Conclusion	57
CHA	PTER FIVE SUMMARY, FINDINGS AND RECOMMENDATION	DNS
5.1	Introduction	
5.2	Summary of the study	
5.3	Findings	59-63
5.4	Recommendations	
5.5	Implications for future research	
5.6	Lessons learnt from this study	
5.7	Conclusion	65-66
REFERENCES		67-80
APPI	ENDICES	
	A Ethical Clearance	81
	B Participants Consent Letter	82-83
	C Principals Consent Letter	84-85
	D Interview Schedule	86-87
	E Turnitin Certificate	88

vi

LIST OF TABLES

Table 3.1	Participants' Profiles	28
Table 4.1	List of Themes	35

LIST OF ACRONYMS

CAPS:	Curriculum Assessment Policy Statement
CK:	Content Knowledge
DHET:	Department of Higher Education and Training
DOE:	Department of Education
ECDL:	European Computer Driving License
FET:	Further Education and Training
HEI:	Higher Education Institutions
HW:	Hardware
ICT:	Information Communication Technology
ICDL:	International Computer Drivers Licence
IW:	Ideological-ware
LMS:	Learning Management System
LO:	Life Orientation
LOs:	Learning Outcomes
NCV:	National Certificate Vocational
NQF:	National Qualifications Framework
NSDSIII:	National Skills Development Strategy III
OBE:	Outcomes Based Education
PCK:	Pedagogical Content Knowledge
PK:	Pedagogical Knowledge
SAGs:	Subject and Assessment Guidelines
SGs:	Subject Guidelines
SW:	Software
TK:	Technological Knowledge

TPACK:	Technological Pedagogical and Content Knowledge
TVET:	Technical Vocational Education and Training
UNESCO:	United Nations Educational, Scientific and Cultural Organisation
UK:	United Kingdom

ACKNOWLEDGEMENT

To God almighty for giving me the strength, wisdom and endurance to complete my study.

The participants, for the time they sacrificed in order to be part of this study.

My supervisor, for her excellent spirit of patience, kindness and hard work.

My campus manager, for allowing me to use the college as the site for research.

My friends and family, for their support.

My children, for their understanding and endurance throughout my studies.

My husband, who is my mentor.

Dedication

This dissertation is dedicated to Kylie-Ann Paul

CHAPTER ONE

INTRODUCTION AND BACKGROUND TO THE STUDY

1.1 INTRODUCTION

This chapter presents a background to the study which incorporates the educational changes that occurred in South Africa after 1994. These educational changes include the National Curriculum Vocational (NCV) curriculum as the new curriculum at Technical Vocational Education and Training (TVET) Colleges. The shift from Curriculum and Assessment Policy design in favour of a learner-centred approach is also emphasised. Focus is on Life Orientation (Computer Skills) lecturers and their experiences in teaching Computer Skills. What their experiences are in teaching using the subject and assessment guidelines in Level Two is also discussed. The background, purpose, research problem, objectives and research questions of the study are discussed. An organisation of subsequent chapters covered in the study is also provided.

1.2 BACKGROUND

In South Africa, the post-1994 dispensation ushered in several changes in the area of curriculum implementation. The National Department of Education was under pressure to provide a curriculum that would address the past imbalances in education (Makgato & Mbanguta, 2002). The Department of Education and the Department of Labour joined together to rectify the skills shortage and 'brain drain' that was evident (Gamble, 2003). In 1994, Outcomes-Based Education (OBE) was introduced as the curriculum for teaching and learning (Makgato & Mbanguta, 2002). The importance of providing education for all was also stressed (Makgato & Mbanguta, 2002). The implementation of the National Certificate Vocational (NCV) curriculum began in 2007. The implementation led to various disruptions, caused by, among other things, lecturers not having adequate training at Technical Vocational Educational and Training Colleges. NCV was 'here to stay' as DHET made clear in the Green Paper: The Department of Higher Education and Training is "committed to NCV" (DHET,

2012, p. 22). In the Green Paper for Higher Education and Training, the Department of Higher Education and Training laid out its aim for the TVET colleges to be "the key area of focus for expansion" of education and training in the country (DHET, 2012, p. 23).

One of the areas of focus of the Department of Higher Education and Training was to form a partnership between industry and TVET Colleges. In essence, this partnership implied bringing the world of work to students during their education (Keeting, 1995). Private sector involvement in teaching and learning provides an additional value for education (Kerre, 2010). This partnership was in alignment with the vision of enhancing South Africa's economy with the skills development of its workforce as mentioned in the National Skills Development Strategy III (NSDSIII). NCV is an outcomes-based approach to education. Hence, it is suggested in the Subject Guidelines (SGs) document that the curriculum is learner-centred. According to the Subject Guidelines (SGs) document for Life Orientation, the lecturer must be suitably qualified in order to effectively teach Life Orientation that is learner-centred.

TVET Colleges serve a dual function of offering both post-matric vocational training (Report 191), as well as the National Certificate Vocational curriculum. The lecturers' experiences are therefore, diverse in nature, depending on which curriculum is taught by the lecturer. NCV is equivalent to grades 10, 11 and 12 at schools (FET phase). These grades are given the designations National Qualifications Framework (NQF) Level 2, 3 and 4 respectively at Colleges. Moreover, this shift from lecturer (teaching post-matric) to 'school teacher' complicated lecturers' everyday experiences as most were not prepared for these challenges (Wedekind, 2016). When teaching the curriculum (Report 191), a trade qualification and industry experience was required at inception. With the new role of 'school teacher', a teaching qualification was necessary. For this reason, many lecturers are now compelled to study towards a pedagogic qualification.

1.3 PURPOSE OF THE STUDY

The purpose of the study was to explore Level 2 Life Orientation lecturers' experiences in teaching Computer Skills at a TVET College. Computer Skills is a component of the National Certificate Vocational (NCV) which was implemented in 2007, as mentioned earlier. This new curriculum was implemented to address the shortage of skills in the labour market and industry. The study explores lecturers' experiences that resulted from the many changes brought about by the NCV curriculum. In addition, the study focuses on lecturers who taught Level 2 Computer Skills, as this is the first year of the three-year programme. Teachers' experiences are very important when dealing with curriculum since creating professional teachers is likely to facilitate the implementation of the curriculum as planned. Teaching resources like the availability of trained or qualified educators are just some of the issues highlighted by Ndjabili (2008). In the teaching of the Computer Skills component of Life Orientation, a trained lecturer is essential.

1.4 RESEARCH PROBLEM

The NCV curriculum includes three mandatory subjects which are similar to the school curriculum, namely; Mathematical Literacy or Mathematics, English and Life Skills (Life Orientation). These fundamental courses are core to all specialisations and are "generic rather than specialised" (Wedekind & Watson, 2016, p. 78). An important issue in relation to lecturer identity is highlighted, as lecturers of Fundamental courses are likely to have a schooling qualification rather than subject specific knowledge. Furthermore, by schooling qualification, reference is made to a teaching qualification that would equip educators in their instruction of the learning area. This study revealed that lecturers of Life Orientation did not possess qualifications that equipped them to teach the subject of Computer Skills; yet according to the Subject Guidelines (SGs) document, the lecturers must be suitably qualified to teach Life Orientation (DoE, 2013b).

It is significant for this study to explore how a township college Computer Skills lecturers experience the teaching of the Computer Skills component of Life Orientation (LO). It is suggested that "ICT lecturer must be computer literate and have an advanced knowledge of the following programmes: Word processing, Spreadsheets, Presentation and knowledge of the Internet and e-mail" (DoE, 2013b, p.13). In numerous countries like Germany, the competence assessment or diagnostic testing is far ahead, and allows for comparison of lecturers' competencies (SAQA, 2016). Tools that facilitate comparing strengths and weaknesses in lecturers' content knowledge, are critical for South African (Rauner, 2005). Reviewed literature is silent on computer studies as a College-based practical component incorporating NCV. This study seeks to bridge that gap. It seeks to understand the teaching of Computer Skills at Level 2, as experienced by lecturers, as this is the first level at which all NCV students are required to take Life Orientation as a compulsory fundamental subject.

1. 5 RESEARCH QUESTIONS

The study sought to answer three research questions:

1.5.1 What are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

1.5.2 Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

1.5.3 How do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching?

1.6 ORGANISATION OF CHAPTERS

The study is divided into five chapters as indicated below:

Chapter 1: Introduction and background to the study

Chapter one gave an introduction to the research that was conducted. A discussion on OBE, the TVET sector, the NCV programme and Life Orientation

was made. The background, purpose, research problem, and research questions of the study were also discussed.

Chapter 2: Review of related literature and theoretical framework

Chapter two discusses the relevant literature that was reviewed in relation to the study. The study explores lecturers' experiences of teaching Level 2 Life Orientation Computer Skills at a TVET College. This chapter begins with a background of a literature review, followed by a discussion of the teaching of Computer Skills internationally and then nationally. A discussion is then made on TVET education under the heading; learner-centred, life-long learning and authentic learning. In conclusion the theoretical framework guiding this study, that is, the Technological, Pedagogical and Content Knowledge (TPACK), is also discussed.

Chapter 3: Research methodology and design

Chapter three discusses the interpretive paradigm around which the research is centred. The chapter then provides a description of the research design and approach that was used for the study. The case study research methodology is also described in this chapter. Following that is a discussion on the data generation using the semi-structured interviews. Selection of participants and the data analysis process are also described. Ethical considerations, trustworthiness and credibility, transferability and dependability as well as the limitations and delimitations of the study are also discussed.

Chapter 4: Data presentation and analysis

Chapter four gives a description of the data sources. The chapter also provides a detailed analysis of the data which is presented, in the form of themes as the study adopted a thematic analysis.

Chapter 5: Summary, findings and recommendations

Chapter five concludes the study by providing a summary of the study, a description of the findings, and recommendations to inform future research.

1.7 CONCLUSION

This chapter provided an introduction and background of the study. The purpose of the study was outlined. Following this, the research problem, objectives and research questions were discussed. The study attempted to answer three research questions, namely: What are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum? Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum? How do lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV stills in the NCV curriculum? How do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching? Finally, the organisation of the chapters was discussed. The next chapter presents the related literature reviewed, and the theoretical framework for this study.

CHAPTERTWO

REVIEW OF RELATED LITERATURE AND THEORETICAL FRAMEWORK

2.1 INTRODUCTION

Technical Vocational Education and Training lecturers offer vocational and academic subjects to young and old people. This chapter discusses the related literature of the study under the following headings: Teaching of Computer Skills - international teaching, national teaching; and TVET education - learner-centred, life-long learning, authentic learning. The theoretical framework used in this study is also discussed. The Technological, Pedagogical and Content Knowledge (TPACK) is used as a framework for the study, encompassing technological knowledge (TK), pedagogical knowledge (PK) and content knowledge (CK). TPACK proved to be the most relevant framework as it presents the components of knowledge that are specific to lecturers of Computer Skills.

2.2 TEACHING OF COMPUTER SKILLS

2.2.1 International teaching of computer skills

The issue of teaching Computer Skills in TVET Colleges is a critical phenomenon for discussion. The literature in more developed countries like South Korea is reviewed as well as that in developing countries like Bangladesh. Other countries like Malaysia, United Kingdom, New Zealand, Australia and Germany's literature is also reviewed in this study.

The computer is an imperative device for teaching and learning in Malaysia. The Minister of Education initiated a project to allow students to gain entry into the world of work (Saud, Rajuddin, Ismail, Nordin, Minghat, Subari, Amin, Hamid & Arsat, 2000). The use of computer technology in Computer Skills was the main reason for the project. Information processing devices became known devices used for teaching in skills based institution (Saud et al., 2000). Computer Skills and computer technology application are widely used in schools and tertiary education institutions in Malaysia. According to Fabry and

Higgs (1997), the major hindrance to implementing computers in teaching and learning consists of; aversion to change, educators' negative attitudes and poor instructional training. In Malaysia, the country has experienced similar constraints as with other countries such as poor vision/rationale for computer use, barriers to training and insufficient support of educators (Saud et al., 2000).

In a study in Korea, Park and Sung (2013) highlight some of the barriers to implementing technology in teaching when they explored how educators handled curriculum change and the support that is needed to effect change. Their findings were that educators were resistant to change and that ultimately had adverse implications on the incorporation of the change in curriculum. The barriers highlighted by Park and Sung (2013) included: lack of mentoring programmes to support educator professional development, failure to work through challenges with colleagues, and cultural and contextual challenges.

A vast number of TVET educators in Bangladesh are motivated to use Computer Skills in education (Raihan & Shamin, 2013). As a developing country, Bangladesh has experienced many constraints in implementing Computer Skills in the education system. These constraints include: lack of computers for students, insufficient knowledge of Computer Skills by educators caused by the low incentive for educators to undertake Computer Skills training, and constant electricity load shedding.

A study in the UK showed that the main activities used in teaching and learning were e-mailing, word processing and searching the internet (Conole, de Laat, Dillon & Dary, 2006). The European Computer Driving License (ECDL) is one of the qualifications from the British Computer Society which is accepted internationally for its credibility. Hence, emphasis by Rickaby (2007, p. 10), that:

The European Computer Driver's License is developed to facilitate a foundation of competence in the skill of information technology and managing of a personal computer. This incorporates the use of data bases, the use of the Internet, presentation and e-mail.

This would then imply that anyone completing the ECDL course would be equipped in all areas of Microsoft Office. Furthermore, it must be noted that the ECDL and the International Computer Drivers Licence (ICDL) are the same. The European Computer Driver's License is simply the same ICDL which the British Computer Society have now laid claim to. According to Rickaby (2007), the ECDL course would represent a benchmark for equipping teachers with the necessary skills for teaching Microsoft Office. As ECDL is an internationally recognised qualification, it would provide a benchmark for skills measurement across the globe. This recognition, would further mean that educators across the globe can be assured that the standard of delivery is the same based on the above mentioned global benchmarking (ECDL Foundation, 2011).

Australia was always in the forefront regarding higher education specialising on the preparation and mentoring of educators. The consequence of not providing the required support would be lack of Computer Skills knowledge (Johnson, 2012). Institutions can learn from Australia for supporting educators in Higher Education in South Africa.

Tools that facilitate comparing strengths and weaknesses in lecturers' content knowledge are critical for South African (Rauner, 2005). In countries such as Germany, a form of diagnostic testing of lecturers' knowledge is way advanced compared to South Africa, and allows for comparison of lecturers' competencies with regard to content knowledge.

2.2.2 National teaching of computer skills

Studies have noted that technologies, like computers, provide rich educational experiences (Zirkle, 2002). Hence, many scholars propose using computers in education (Day, Raven & Newman, 1998). These scholars discovered that students who were taught via computer technology performed better compared to students whose education experience was only based on the traditional method of teaching.

The literature reviewed echoes Brown and Czerniewicz's (2008) report confirming the use of Computer Skills for teaching and learning in South African HEIs, with some institutions still lagging behind. Regarding HEIs in South Africa, the most frequently used Computer skill was finding information (Internet) and the use of word processing (Ms Word). The literature reviewed is consistent with other findings like those in the United Kingdom regarding the main activities undertaken in Computer Skills. The SAGs recommends that the "ICT lecturer must be computer literate and have an advanced knowledge of the following programmes: Word processing, Spreadsheets, Presentation and knowledge of the Internet and e-mail" (DoE, 2013, p.13).

In a study conducted by Ntshaba (2012) in King Williams Town, educators experienced a lack of confidence in teaching due to insufficient content knowledge. Jones and Moreland (2004) conclude that for educators to teach successfully, they should have technological, pedagogical and content knowledge of the subject taught. Hence, this would imply that the lack of content knowledge will result in educators only focusing on learning areas that are known to them, to the detriment of students (Pool, Reitsma and Mentz, 2013). The result of lecturers' scarce technological, pedagogical and content knowledge would suggest that a benchmark system should be used whereby lecturers' progression in Computer Skills is recorded. In South African literature, there is a gap concerning the competency assessment for lecturers. These tools to employing a benchmark system are evident and in use in countries like Germany. Hence, the strong recommendation by bench markers, for lecturers to complete the International Computer Drivers Licence (ICDL), as the ICDL will provide South Africans with a form of diagnostic testing of lecturers' competencies in Computer Skills.

There are five components that make up the complete ICDL course. These components are word processing, presentations, spread sheets, internet and e-mail. The ICDL course would indicate that the lecturers of Computer Skills have an up-dated and relevant qualification (Sami, 2007). Furthermore, the qualification is internationally recognised and provides a good foundation for Computer Skills which assists lecturers to be confident and competent in their

computers content knowledge (CK) (Sami, 2007). This would help lecturers in delivering their instruction and pedagogical knowledge (PK) of the NCV curriculum competently.

Training lecturers to use computer aided devices in teaching delivery may assist students to embrace technology. Training is essential as it gives teachers the advantage to spearhead the curriculum (Hough & Neuland, 2014). Students enjoy the use of computers and therefore computers can be used as a successful teaching tool, for example Blogs (Majid, 2014). Studies have indicated that students are knowledgeable about the use of computers and also the dangers associated with the use of computers.

2.3 TVET EDUCATION (LEARNER-CENTRED)

Ngubane-Mokiwe (2014), suggested that TVET education need to respond to the labour market and centre on the student. The focus on the learner rather than the educator is in line with the suggestion for TVET Colleges to move away from an out-dated understanding of their purpose, to becoming more resilient and adaptable (Kraak & Hall, 1999). South Africa can be likened to Australia that there are hindrances to TVET Colleges' success, such as, insufficient resources; lack of teaching approaches and inappropriate TVET skills of lecturers (Goldney, Murphy, Fien & Kent, 2007). The issues highlighted in Goldney et al. (2007) report are parallel to the South African landscape where educators are finding difficulty in moving from traditional teaching approaches to more contemporary approaches. The learner-centred approach is used in NCV because it incorporates the three important elements of learning outcomes (Khoza, 2013bQ). These three elements are made up of intended, implemented and attained outcomes. The intended outcomes constitute the Learning outcomes (LOs) in the Subject and Assessment Guidelines (SAGs) of Life Orientation. The SAGs is also a policy document for NCV. It is what is planned before learning begins. The implemented outcomes would be what occurs during learning. Lastly, the attained outcomes would be observed from students and assessed after learning has occurred. Hence, it suggests that learning outcomes are meant to be learner-centred rather than teacher-centred.

Although, it is suggested by Anderson and Elloumi (2004) that all three approaches namely; teacher-centred, content-centred and learner-centred approaches should be incorporated, none of them should be used in isolation. If the lecturer wants a suitable approach for presentation, then the teacher-centred approach would suffice. Alternatively, if the lecturer wants to grade content, then the content-centred approach would do. If the lecturer wants to contextualise learning then the learner-centred approach would be adequate.

2.4 TVET EDUCATION (LIFE-LONG LEARNING)

The then Minister of Education Naledi Pandor noted in the FET Colleges Bill (RSA, 2006 p.1) that:

We have redesigned the College sector so that from next year they will be able to offer intermediate and high-level skills to students from the age of 16 to mature adults. They will be able to give effect to our long held ideal of providing lifelong learning. To fulfil this goal, Colleges have to be a different type of institution to the FET schools.

Lifelong learning also adds to sustainable living which is a SAGs requirement in the TVET institutions NCV curriculum. Vocational education and training must provide training to people to allow them to be self-sustaining (Powell, 2012b). Hence, TVET institutions, via LO Computer Skills, should be the training grounds that will ensure students employability in industry. Some researchers such as King (2012) and McGrath (2012) question the relevance in the workplace of TVET institutions considering that these institutions have a long existence, but very little success in skills training for student employability. A solution is offered by McGrath (2012) who recommends "reimagining the purpose" (p. 36) by TVET institutions for education that provides employability. Bonvin and Galster (2010) posit, that expanding employability for people, results in them being more sustainable employability, and training in practical skills is pivotal.

Training is essential because it helps educators to be ahead of the curriculum (Hough & Neuland, 2014). TVET institutions provide an education that caters for practical skills, and in this study it is Computer Skills (Finch & Crunkilton 1999). One of the key researchers who are instrumental in curriculum studies is Jeanne Gamble. She discusses the curriculum issues at a theoretical level to better understand knowledge in curriculum studies (Gamble, 2006). In addition, she explores the knowledge required via the curriculum for employment of students in the workplace and the tensions between theory and practice in the curriculum. Gamble argues for the curriculum's need to focus on employability. In conclusion, she recommends that alternative forms of knowledge needed for employability help when rethinking the curriculum and for those who should be teaching the curriculum.

Most of these studies have sought to help TVET institutions to provide practical skills, like Computer Skills. What is clearly lacking in these studies are the voices of lecturers who teach practical skills, like Computer Skills. There, is therefore need for this present study, which explored lecturers' experiences of teaching Computer Skills at a TVET College. The NCV focus is on lifelong learning that gives priority to the skills and demands of the South African economy with the goal to provide further education and training and to ensure employment and productivity (DoE, 2007).

2.5 TVET EDUCATION (AUTHENTIC LEARNING)

Learning with technology incorporates principles of authentic learning (Reeves, Herrington, & Oliver, 2004). There are two components in the incorporation of computer technology in learning. First, is the impact of technology to alter pedagogical practices. Second, is the incorporation of computers as facilitators of learning.

Technology should contribute to cultural and social interactions to allow for positive learning outcomes. The National Curriculum Vocational (NCV) education is characterised by a broad approach to education and training with a

change from a theoretical curriculum to a curriculum in which both theory and practice are combined. Authentic learning is stressed in the Subject Guideline which states that the student must encounter an authentic community (DoE, 2013). Authentic learning allows learners to connect their learning area with community issues, which also enhances the practicality of the curriculum. For Life Orientation to facilitate learning, it should use real activities as "Authentic activities have real-world relevance" (Herrington, Reeves, & Oliver, 2004, p. 3).

Theron's (2008) study was based on students, especially those who lived in townships. The focus of his study was on the LO teachers' knowledge of the students' backgrounds and how that influenced the manner in which the teacher interacted with the students. The study's findings were that the subject Life Orientation requires that students should be taught in a holistic way. This concurred with the Subject Guidelines (DoE, 2013b) which stated that students must be exposed to authentic issues that will create an authentic platform within which learning can take place.

Amory (2010) suggests that learning environments must provide collaborative opportunities that support teaching and learning. A shift needs to be made from behaviourist and instructionist approach, which perpetuate passive behaviour, but rather to a leaner-centred approach which produces individuals and communities that are active participants.

2.6 THEORETICAL FRAMEWORK

2.6.1 TPACK framework (technological, pedagogical and content knowledge)

The TPACK framework is an extension of Shulman's formulation of "pedagogical content knowledge" (PCK). Historically, teacher knowledge has focused on content knowledge of the educator (Shulman, 1986). With time, teaching knowledge has moved to pedagogical knowledge to the detriment of content knowledge (Ball & McDiarmid, 1990). Shulman treated pedagogical and content knowledge as two exclusively separate components (Shulman, 1987). The result of separating the two components was a focus on only one of the

components in teacher education. However, Shulman advocates for an approach that includes both pedagogical and content knowledge. This explains the introduction of the two knowledges based on their inseparable relationship. Although Shulman did not highlight technology as a component and its relationship to pedagogy and content, he did not deem it irrelevant. When he first introduced his notion of PCK, issues around technology had not gained momentum as they have today. Much has changed and new technologies have come to the front line of educational discourses. Technology has changed the arena of the classroom and continues to do so. The new technologies include both hardware and software (Mishra & Koehler, 2006).

In their five-year study, Mishra and Koehler (2006), focused on lecturer development and discipline development in higher education. Their study has enlightened theory and practice, with an added advantage of bringing together theory and practice. The target group consisted of educators and practitioners, as well as researchers and theoreticians. They debate, briefly, how thought, through pedagogy implementations, need a concise type of knowledge that they refer to as Technological Pedagogical and Content Knowledge (TPACK). Therefore, Mishra and Koehler (2006) highlight the relationship that exists between the three main parts of the learning environment namely; technology, pedagogy and content.

The lecturers' experiences of teaching computer skills are displayed by using the Technological, Pedagogical and Content Knowledge (TPACK) framework. According to studies teaching Computer Skills requires appropriate theories (Khoza, 2013b). Hence, teaching is not centred on technology but around the teaching approaches used (Amory, 2010). These teaching approaches are used to combine the course content with suitable technology (Mishra & Koehler, 2006). For this study TPACK is the appropriate framework as it is relevant for teachers who teach Computer Skills. The teaching of Computer Skills (technology) needs relevant theory, hence the use of TPACK.

Frameworks and theories assist us in understanding the world. The frameworks provide us with theories with which to understand phenomena closely. The

TPACK framework allows the researcher to explore the teaching of Computer Skills, which includes technology, pedagogy and content knowledge. The TPACK framework helps to combine all three components. TPACK protects against the teaching of Computer Skills (technology) separately, combines all three areas of technology, pedagogy and content knowledge. Mishra and Koehler's (2006) TPACK framework is not a new notion, as other scholars have also argued in favour of TPACK. Studies are in agreement that technology cannot be separate from pedagogy and content (Hughes, 2005; Keating & Evans, 2001; Lundeberg, Bergland, Klyczek, & Hoffman, 2003; Margerum-Leys & Marx, 2002; Neiss, 2005; Zhao, 2003). What is different about Mishra and Koehler's (2006) approach is the precise nature of the link between technology, pedagogy and content.

2.6.1.1 Technological knowledge

Technological knowledge (TK) is about the use of overhead projectors, screens and any modern device (computers) used in the classroom. In addition, TK includes the implementation of electronic computers and computer hardware and software to manipulate, transmit, secure and restore data (Rickaby, 2007). Atkinson and Mckay (2007) also suggest that technology employs electronic technology that generates, stores and processes data.

A study by Khoza (2013a) identifies three types of resources (materials) used in the teaching and learning process, namely; hardware, software and ideological ware. Hardware would include any object, tool and machine used in teaching. Hardman (2008, p. 25) posits that "tools mediate thought during the interaction between the subject and the context within an activity". Software incorporates any material used together with tools to display information. Ideological ware are "things that we cannot see and touch in education, such as theories and others" (Khoza, 2015a, p.1). These studies indicate that teachers use all three as a medium to disperse knowledge, but it is within the ideological-ware that learning takes place. Learning is not about teaching and learning resources where students learn from learning resources, but about ideology where learners learn with resources (Amory, 2010). Other resources as stipulated by

the Subject Guidelines document (SGs) are well-equipped classrooms, media centres, and a suitably equipped computer room. Under consumables are; files for projects, printing paper, cartridges for printers. These selected resources, as outlined by the Subject Guidelines document, are both relevant and practical to the teaching of Life Orientation in order to ensure the sustainability of the curriculum.

As technologies are constantly changing, the educator is required to keep abreast of these changes in Technological knowledge. This calls for educators to rethink their pedagogical knowledge (Peruski & Mishra, 2004; Wallace, 2004). In order for educators to teach, they must find ways to link educational instruction with sufficient computer technology, mainly the internet. Hence, technological and pedagogical knowledge are thus far inseparable in TPACK.

2.6.1.2 Pedagogical knowledge

Pedagogical knowledge (PK) incorporates the teaching method and practices of conveying the content knowledge to students. Hence, this will indicate the lecturers' reasons for their teaching. (Why are lecturers teaching?) Berkvins et al. (2014) indicate that teaching reasons are categorised into pedagogical/personal reasons, societal/social skills development reasons, and content/metacognition development reasons. Personal reasons would encompass knowledge that is unique to each lecturer and which holds personal significance (Schiro, 2013). TVET College lecturers work with two curricula which in itself can be very confusing when trying to teach. As in any given working day, lecturers have to switch from teaching one curriculum NCV which is driven by learning outcomes to another curricula Report 191 which is driven by subject content.

Goals are made up of aims, objectives and outcomes. A clear distinction needs to be made between aims, objectives and outcomes. The aims, as suggested by the Subject Guidelines document, are drawn for all teaching intentions and are long-term goals (DoE, 2007). The teacher's intentions in a learning area are referred to as the aims of a learning area (Kennedy, Hyland, & Ryan, 2006).

These are broad general statements which can be seen as desires for teaching. It is imperative that the aims be documented from the teacher's perspective (Kennedy et al, 2006). Objectives are not as broad as aims, but rather, are specific intentions of the lecturer. These objectives are also precise statements of education, and can be noted as the action of educating. Outcomes are learner-centred and TVET education focus is around the learner-centred approach.

The National Certificate (Vocational) Life Orientation curriculum is an outcomesbased approach to education. A learning outcome (LO) is a statement of that which the student is expected to grasp at the end of learning. Every module has its separate LOs that are prescribed in the Subject and Assessment Guidelines (SAGs). Educators use learning outcomes to teach their modules (Moon, 2002). Moon (2002) also states that learning outcomes rather than objectives facilitate learning. Objectives gravitate around the lecturers' intention rather than the students' intentions. The consequence of lecturers using learning outcomes to teach their modules is that learning outcomes can be observable and assessed by lecturers.

In a study on outcomes of computer technology, Khoza (2001) suggests the need for linking learning outcomes to assessment methods. Assessments can never be over-emphasised in the learning and teaching process. As far as the learner is concerned, "the assessment is the curriculum" (Ramsden, 2003). The tools for assessing students as outlined in the Assessment Guidelines document of Life Orientation are rubric, checklists, task lists and rating scales (DoE, 2013a).

Assessing whether lecturers have the required pedagogical knowledge will depend on their qualifications. The focus should be on lecturers' pedagogic qualifications. Literature suggests that there is a lack of pedagogy on how to teach Computer Skills and for creating lifelong learning. The literature reviewed shows a gap in understanding the pedagogy that is applied in the technical and vocational institutions. The pedagogy is referred by Shulman (2005) as the

"signature pedagogies". Shulman (2005) refers to the common pedagogy of technical and vocational institutions.

2.6.1.3 Content knowledge of curriculum

The content to be taught is covered in content knowledge (CK). This would include theories, facts and concepts concerning their learning area. Educators who do not have the required CK can erroneously misguide students in their teaching (Ball & McDiarmid, 1990). In view of the fact that lecturers mainly work with knowledge, one of the main areas that needs to be researched would be the nature of content knowledge.

Taylor (2010) reports that some teachers would not pass the assessment for which they are supposed to be assessing students on. It would then suggest that teachers' content knowledge is limited. Teaching Computer Skills requires a competent lecturer in the computer field, and in possession of an updated computer qualification such as in the ICDL certificate (DoE, 2013b). If adequate training is not received in Computer Skills by educators, Life Orientation may not reach its goal of equipping students with Computer Skills. In LO, the assessment guideline makes use of formative assessment as inclusive of both formative and summative assessment. Learning outcomes should be drawn up to include both summative and formative assessment methods. A study is needed to ascertain precisely what lecturers' experiences in teaching Computer Skills are.

2.7 CONCLUSION

This chapter provided an overview of the related literature of the study. This chapter began with a background of a literature review, followed by a discussion of the teaching of Computer Skills internationally and then nationally. A discussion was then presented on TVET education under the following headings; learner-centred, life-long learning and authentic learning. In

conclusion the theoretical framework of the research study; the Technological, Pedagogical and Content Knowledge (TPACK) was discussed. The above review demonstrates clearly the need for the study. The need for the study is the paucity of research on experiences of Level 2 Life Orientation Computer Skills lecturers. The following chapter presents the research and methodology design for the study.

CHAPTER THREE

RESEARCH METHODOLOGY

3.1 INTRODUCTION

In the previous chapter, a review of both international and national literature on teaching of Computer Skills was presented. This chapter provides an in-depth description of the methodology which was incorporated in the study. It includes a description of the research paradigm, design and research approach that were adopted in this study. A detailed view of methods employed to generate data is also discussed. In conclusion, a discussion on the analysis procedures and methods used is included. The ethical considerations of this research are also included in the conclusion of this chapter.

3.2 RESEARCH DESIGN

Research design is the precise and on-target employment of research methods where various tools and procedures are incorporated to execute different tasks (Mouton, 2001). The section begins by discussing the paradigm used in this study which is the interpretivist paradigm. This study also employed a qualitative research approach, within this paradigm. This section also discusses how I conducted the research and how the various stages of the research were approached and completed.

3.2.1 Interpretive paradigm

The way we view the world is explored by a paradigm. Punch (2013) observes that paradigms employ various models, ideals and values that researchers have in agreed on concerning research. A paradigm gives the impression that there is a chain of thoughts or images one uses to see the world through (Guba & Lincoln, 1994). The interpretive paradigm was used to underpin this research. In using this paradigm, the researcher's purpose was to understand and describe meaningful social action about how people comprehend their circumstances

and experiences (Christian, Bertram, & Land, 2010). In using the interpretive paradigm, I wanted to understand experiences of Level 2 lecturers in teaching the Computer Skills component in Life Orientation. The paradigm sees truth as multiple and allows no room for generalisation of findings as the truth "recognises multiple interpretations as equally valid" (Christian et al., 2010, p. 23). I therefore, used this paradigm to give a voice to lecturers of Life Orientation on the experiences that they have in the classroom. The purpose of the study was to explore Level 2 Life Orientation lecturers' experiences in teaching Computer Skills at a TVET College.

3.2.2 Qualitative approach

This research study is qualitative in design. Qualitative research takes place in the normal settings of the participants, to ensure that the data generated would be as authentic as possible (Merriam, 1998). McMillan and Schumacher (2010) assert that in the qualitative approach, data is generated on naturally occurring issues. I chose lecturers as participants in order to ascertain their perspectives and experiences because qualitative studies are about understanding the reality from the views of those living in it (Patton, 2005). In this study, I made use of interviews to understand the participants' experiences which varied form one lecturer to another. These experiences were based on their naturally occurring context, which is the classroom within the college, so as to construct meaning from the lived experiences of lecturers.

The advantage of qualitative research is that the participants are able to bring various life experiences and conceptions into the study (Newby, 2010). Evidence in the form of data generated from interviews of the participants' realities can be obtained. The conception of the participants' nature of reality cannot be generated through quantitative research, as it lacks the personal aspect (Essack, 2011). In the study, the personal nature of qualitative research allowed the participants to provide thick descriptions of the phenomena when interviewed. Another advantage of using the qualitative approach was that it enabled me to get a deeper meaning of what experiences lecturers had in the

classroom, providing rich quality data. Hence, the qualitative nature of the study provided a deeper insight, as it explored how Life Orientation lecturers taught in the classroom. In a qualitative approach, the researcher aims to interpret the data by using evidence from the data and literature (Henning et al., 2005).

3.3 RESEARCH METHODOLOGY

This section describes the methodology used in this study. A case study was the chosen methodology for this study. This section provides an outline on the chosen methodology.

3.3.1 Case study

In this study, I wanted to understand why LO lecturers have the experiences that they do in teaching Computer Skills, hence a case study was the appropriate methodology. A case study can be defined as a wide expedition of a certain case in order to obtain data (Rule & John, 2011). The case study is defined by the researcher's fascination in a particular case and not by a methodological choice. Yin (2014) asserts that case studies are used to study the 'case' in its real world context. In incorporating a case study, it was interesting for me as the researcher to ascertain how lecturers' experiences influence their teaching of LO Computer Skills in NCV, as my chosen case. The 'case' was the experiences of four LO Computer Skills lecturers at a TVET College in a township. It was not this study's intention to generalise about the findings, but to obtain in-depth understanding of the LO Computer Skills lecturers' experiences of teaching LO Computer Skills.

According to Cohen, Manion and Morrison (2011), a qualitative case study can be defined as an in-depth, wide description of a case. The case study focuses on experiences rather than results, on exploration as opposed to confirmation. Hence, emphasis is placed on interpretation, meaning and understanding, rather than experimentation and deduction (Cohen, Manion and Morrison, 2011). I chose a case study in order to gain an in-depth understanding of experiences of LO lecturers who taught Computer Skills. Cohen et al. (2011) highlight some of the disadvantages of case studies, such as case studies not being open to cross checking and being subjective, personal and selective. This study had a small sample and hence the data generated and findings were personal, contextual and subjective, and could not be generalised.

3.4. Data generation

This study made use of semi-structured interviews as a method of data generation to answer the following questions:

3.4.1. What are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

3.4.2. Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

3.4. 3. How do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching?

The semi-structured interview method allowed me to obtain data on how lecturers experienced teaching Computer Skills at a TVET college. The data was generated from the interviews which occurred at the college where the lecturers were at ease, as it was a familiar environment for them.

A voice recorder was used to record interviews with participants. Since the participants were informed about confidentiality and anonymity of the study, the recording did not produce any mistrust.

3.4.1 Semi-structured interviews

An interview is a flexible tool of data generation which employs verbal and nonverbal channels of communication to be used (Cohen et al. 2011). I made use of interviews in my research as it is suggested that interviews are one-to-one direct conversations with participants, making use of questions, which are designed to extract detailed responses (Kvale, 1996). Semi-structured interviews allow the researcher to list categories and questions with opportunities for open-ended discussion of items.

Semi-structured interviews are those centered around a specific area, while still allowing flexibility, in depth and scope (Dicicco-Bloom & Crabtree, 2006). The use of open-ended questions allowed me the opportunity to probe ideas and views from my participants (see Appendix D Semi-structured interview schedule). Probing is a skill used by the researcher to discover alternative avenues that were not first thought of by the interviewer or interviewee (Gray, 2004).

One advantage of semi-structured interviews is that they allow for open-ended questions as opposed to structured interviews which incorporate mainly closed questions (Saunders, 2000). Another advantage is that they allow the researcher to probe and explore items to get above the normal responses of structured interviews. As an interviewer, I was able to obtain additional data, through probing. Furthermore, probing was not the same for all participants since every participant had a different experience.

When conducting the semi-structured interviews, I had much more liberty than that of a researcher conducting a structured interview in that I did not have to follow a detailed research guide (Kanjornboon, 2004). Due to the liberty that I had during the interviews, I was able to get different perspectives from participants on the experiences they have in teaching Level 2 LO Computer Skills. These perspectives gave me deeper insight into participants' experiences. This study concurs with Polzer's (2007) idea, he terms concerning research 'from below', meaning getting knowledge on the experiences occurred from the participants' points of view. The interviews were face-to-face and therefore, non-verbal communication by the participants was easily identifiable. Furthermore, I was able to understand the lecturers' views about their teaching experiences at a TVET College. The evidence was provided while communicating with participants and from facial expressions by participants. Since some views were not clearly put across by participants, the non-verbal gestures, which helped in making sense of the data. Creswell (2012) states

that interview data may be deceptive and provide viewpoints anticipated by the researcher.

Critical conversations on lecturers' teaching experiences provided important information for my study as they expressed important issues they experienced. Emphasis was placed on the conversations rather than the questions. Moreover, questions unfolded in a conversational manner resulting in participants expressing important issues they experienced (Longhurst, 2003). The critical conversations, achieved through open-ended questions and through probing, allowed me to establish good relationships with my participants. Burdick and Sandlin (2010) state that the use of critical conversations create inter-personal relationships. These inter-personal relationships are a result of the in-depth information provided by participants.

Disadvantages of semi-structured interviews are that they are commonly seen as 'easier' (Wengraf, 2000, p.1). Novice researchers, like me are of the view that there is not much planning required for semi-structured interviews. This could not be further from the truth, as I spent much time planning my interviews. Although they are semi-structured, they must be prepared and planned. In comparison to structured interviews, semi-structured interviews need much work to be done in order to be successful. Firstly, they are much longer to complete compared to structured interviews which are fairly quick to complete. Secondly, more creativity and perseverance is needed by the interviewer during the interview session, as semi-structured interviews allow for several themes and questions with opportunities for open-ended discussion of items (Saunders, 2000). Hence, the valuable themes I derived from discussions, compensated for the time needed for creativity and perseverance. Lastly, much more time for transcription is needed after the interview session (Johnson & Onwugbuzie, 2007) as semi-structured interviews are able to generate more data than a structured interview can (Maree, 2007). Moreover, I was able to endure the tediousness of semi-structured interviews as I obtained information rich data that would not have been possible with structured interviews. Also, the researcher needs to guard against obtaining information that is not relevant to the topic researched. The ability to distinguish what is relevant and irrelevant is
of vital importance. I was able to successfully sift through data and find relevant information for the study. Hence, the challenges encountered had little impact on the study.

3.5 SELECTION OF PARTICIPANTS

3.5.1 Sampling

Sampling is the procedure of selecting a manageable number of participants to be part of the study (Dawson 2006). The number of participants chosen for this study was based on the rationale that in the qualitative approach, researchers are concerned with quality and not quantity. It is not the number of participants that is important, but rather the rich information that can be obtained from participants.

The four most accessible Level 2 Computer Skills Life Orientation lecturers were conveniently selected for this study. It is suggested by Bauer and Gaskell (2003) that qualitative research steers clear of large numbers but rather utilises small numbers that produce data of quality. My participants were given pseudonyms. Cohen, Manion and Morrison (2011) note that purposeful sampling is a method of choosing participants that are deemed to be similar to the sample researched. McMillan and Schumacher (2010) concur when they state that 'knowledgeable' participants about the phenomena must be selected. Therefore, for this study, I selected four lecturers who were familiar with phenomenon researched and were from the same environment. Furthermore, convenient sampling allowed me to receive the required data and participants were willing to participate as they were experts in their field.

3.5.2 Participants' profiles

TABLE 3.1 IIIUSITALION OF PARTICIPARTS PROHIES
--

PSEUDONYMS	QUALIFICATION	SUBJECTS TAUGHT	EXPERIENC E	GENDER	AGE	
Mrs Sithole	B.ED. HONS	-LO LEVEL2 (NCV) -Public Finance (Nated)	21 years	FEMALE	44	
Mrs Mpisi	NPDE	-LO LEVEL2 (NCV) -Mathematics (Nated)	7 years	FEMALE	30	
Ms Khan	ACE	-LO LEVEL2 (NCV)	16 years	FEMALE	36	
Mr Guy	PGCE	-LO LEVEL2 (NCV)	12 years	MALE	42	

TABLE 3.1. All four participants taught Life Orientation (Computer Skills) Level 2. Three of the participants were females and one was male. The table shows the varied levels of qualifications that the participants had. At the time of the study, Mrs Sithole had a Bachelor of Education Honours (B.Ed. Hons.), Ms Khan an advanced Certificate in Education (ACE), Mrs Mpisi a National Professional Diploma in Education (NPDE), and Mr Guy a Post Graduate Certificate in Education (PGCE). The youngest lecturer was 30 years old and the oldest was 44 years old.

3.6 DATA ANALYSIS PROCESS

The study used a qualitative method and the data from semi-structured interviews were analysed using themes. Identifying themes has its place in qualitative research. Themes are a starting point in a report of findings from a study (Bazeley, 2009). Any processes that are used to analyse data should be described at length, so as not to leave out any important data (Creswell, 2012).

My data analysis included; reading and rereading of data, coding and identification of themes, and also naming of themes.

To begin with, I had to make sure that all transcripts of interviews were available. The transcription process was very time consuming, evident in the many hours I spent getting the process complete. Following which, I had to carefully read all transcripts and make sense of them. Not all participants were clear as some of their views could not be clearly interpreted. I had to return to participants at a later stage and get them to check their data, so as to see if the data captured their true meaning, in the form of "member-check" (McMillan & Schumacher, 2010, p.3). Going further, I had to reread the data to attach meaning to it in order to distinguish what was relevant and irrelevant. For the purpose of the study, I used thematic analysis to analyse data. According to Rubin and Rubin (2011), analysing data involves identifying themes and patterns in the data. Rubin and Rubin (2011) also argue that data can be analysed in various ways leading to different findings. Guest, MacQueen and Namey (2011) suggest that in thematic analysis one has to go beyond measuring exact statements and words and choose both explicit and implicit views from the data. It occurs when themes are chosen from the data generated. The next step was to code the data and place them into categories. The chosen recurring themes obtained from the data were used to understand the lecturers' experiences in teaching Level 2 Life Orientation Computer Skills. The identified themes were; familiarity and non-familiarity with approaches, availability and unavailability of adequate resources, lecturer training and professional development supported by DHET, teaching experiences, time allocation for teaching and learning, lecturer communication and collaboration, computer skills as a vocational subject, subject specialisation and status of computer skills in comparison to gateway subjects in NCV.

3.7 ETHICAL CONSIDERATIONS

Ethics ensure moral guidelines for researchers providing a method to conduct research in a morally acceptable manner (Struwig, Struwig, & Stead, 2001). In addition, Christian et al. (2010) observe how pivotal ethics is in research as it

assures the protection from any damage that could be incurred by the research. To ensure the participants were protected, I had the participants to sign a consent letter; this ensured that participants were in agreement to take part in my research (see Appendix B). Moreover, I got participants to make an informed decision regarding them wanting to take part in my study or not. I provided an explanation of the purpose of my study which was to explore lecturers' experiences on the teaching of Level 2 Computer Skills at a township TVET college. Providing a purposes for the study was consistent with literature which recommend that the purpose of the study must be fully disclosed to the participants to allow for informed consent (McMillan & Schumacher, 2010). Furthermore, McMillan and Schumacher (2010) suggest that ethics from a moral stance, concerns what is right or wrong. A written and verbal notification (informed consent) was issued to participants ensuring them of anonymity and confidentiality concerning their voluntary participation in the study. To ensure anonymity, as stated above, I used pseudonyms to replace the participants' actual names. According to Creswell (2014), the participants' real names should not be used due to ethical considerations. In addition, participants were informed that they could withdraw at any stage of the study. Confidentiality was assured as all notes, transcripts and audio recordings and any other data was stored in a safe place.

Moreover, permission was also obtained from the principal of the college where the research was done (see Appendix C). The use of the voice recorder as a data generation tool and the purpose of the study were explained to the participants. This resulted in participants agreeing to be recorded. In addition, participants indicated their approval to take part in the study before the research commenced. Approval was also obtained from the university to conduct this study.

3.8 TRUSTWORTHINESS AND CREDIBILITY

Validity can be described as the measures at which the researcher describes the aims of the research (Wilig, 2001). These measures often include numbers which are often not 'validated', in qualitative research. However, in qualitative research, instead of validity; trustworthiness in relation to credibility, transferability and dependability is often used (Punch, 2013). "Trustworthiness is the awareness of the researchers' assumptions, pre-dispositions and influence on the social situation" (McMillan & Schumacher, 2010, p. 374). To ensure trustworthiness of the data, all conversations were audio recorded. I made every attempt not to allow my own "bias" ideas to influence the study. Christian (2010) suggests that trustworthiness also confirms the interpretation of the findings to "reflect the reality and lived experiences of the participant" (p. 49). The aim of a sound research design is to ensure results that are judged to be credible. Credibility deals with the question 'How congruent are the findings with reality?' Therefore, the researcher must promote confidence that accurately records the phenomena under scrutiny. Hence, credibility refers to the extent to which results are similar to reality and are measured to be reasonable and trustworthy (Fortune, Reid & Miller, 2013). To ensure credibility, I returned the transcripts to participants to confirm that it was each participant intended views and voice that was interpreted as a result of being interviewed. This was in line with Conrad and Serlin (2011), who suggest that credibility is observed when the participants confirm the interpretation and construction of the data presented by the researcher.

3.9 TRANSFERABILITY AND DEPENDABILITY

Transferability is the degree to which data can be generalised (Thomas, 2013). "The aim of any study is not to attain generalizability but to understand the phenomenon under investigation though of course the work may perhaps be transferred to similar contexts as the work undertaken" (Thomas, 2013, p. 32). Transferability allowed the readers of a study to relate to other similar studies (Shenton, 2004). I provided information on the number of participants in this study, as this is imperative to ensure transferability. Other information such as the data generation method of semi-structured interviews was also provided and discussed. Furthermore, I provided background data to establish the context of this study. A detailed description of this study was provided to enable readers to have an accurate understanding of information when comparing to other similar studies, so as to promote transferability of my study. I was consistent with the data generation method of semi-structured interviews as the interview transcripts were typed verbatim, thus establishing dependability. More so dependability requires accurate and direct information of the study as derived from the semi-structured interviews. Hence, I generated evidence of data in this study by enclosing direct quotations received from the interviewing sessions to enable readers to assess findings. Struwig, Struwig and Stead (2001, p.169) state that "if raw data are summarised they no longer become original data". Going further, the use of a voice recorder during interviews ensured dependability of the findings by providing evidence in the form of interview transcripts, which could be stored up to a period of five years in a safe place.

3.10 LIMITATIONS OF THE STUDY

I expected that some participants might not be willing to participate in this study as they were my colleagues and perhaps could be afraid to disclose certain information. I assured my participants of the confidentiality in data generation by getting them to sign confidentiality forms (informed consent, see Appendix B). Participants appeared to be at ease and communicated openly with me during interview sessions, hence this fear of disclosing information to a fellow colleague was reduced. The study was also done at the college, where I am presently a lecturer. This could have hindered my research in that I am also teaching at the same college. I had to remind myself to be impartial and not get personally involved in the research. Delimitation is the choices made by myself and the boundaries that I have set for the study. I chose lecturers as participants, as they also are the ideal participants. They teach at the college and hence have firsthand experience of the topic researched.

3.11 CONCLUSION

In this chapter, an exploration of the research design, research paradigm and research methodology used was provided. The study was conducted within the interpretive paradigm that followed a qualitative approach. Semi-structured interviews were the method used for data generation. Confidentiality and anonymity were guaranteed to participants as they were fully informed of the

research prior to commencement of the study. The limitations of the study were highlighted in this chapter to confirm the issues that could impact the credibility of study. Furthermore, the strengths of this study far out weighted the weakness, allowing for the purpose of the study to unfold. The implication for analysis of my study is that I need to collate the huge potentially interesting information and detect meaning of it in the form of themes. Thus, in the next chapter a data analysis and discussion is presented.

CHAPTER FOUR

DATA PRESENTATION AND ANALYSIS

4.1 INTRODUCTION

In the previous chapter a discussion on the methodology adopted in this study was presented. This chapter is a presentation of data and an analysis thereof. The aim of the study was to explore the lecturers' experiences of teaching Level 2 LO (Computer Skills) at a TVET College. The data generated was categorised to answer the following questions:

4.1.1 What are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

4.1.2 Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

4.1.3 How do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching?

4.2 THEMATIC ANALYSIS OF DATA

I made use of thematic analysis in analysing the research data. The Technological, Pedagogical and Content Knowledge (TPACK) framework was used as a guide to analyse data.

The findings are presented thematically, by means of direct quotations, giving meaning to the voices of lecturers and the subsequent discussions to provide 'thick description' (Creswell, 2014, p 41). "There is no single or correct way to analyse and present qualitative data; how one does it should abide by the issue of *fitness for purpose*" (Cohen, Manion, Morrison, 2007, p. 461). For this reason, the researcher must be aware of what the data analysis should accomplish so as to employ the correct method of analysis. The presentation and analysis concerns the data that were generated through the semi-

structured interviews with the participants (see Appendix D). Their responses are categorised so as to determine common patterns (Strauss & Corbin, 1998).

THEME NUMBER	THEME NAME
Theme One	Familiarity and Non-familiarity with
	Teaching Approaches
Theme Two	Availability and unavailability of
	Adequate Resources
Theme Three	Lecturer Training and Professional
	Development Support By Department
	of Higher Education and Training
Theme Four	Experiences of Teaching
Theme Five	Time allocation for teaching and
	learning
Theme Six	Lecturer Communication and
	Collaboration
Theme Seven	Computer skills as a Vocational
	Subject
Theme Eight	Subject specialisation
Theme Nine	Status of Computer Skills in
	comparison to gateway
	subjects in NCV

TABLE 4.1 List of themes

4.2.1 Theme One: Familiarity and Non-familiarity with Teaching Approaches

The learner-centred approach is used in NCV because it incorporates the three important elements of learning outcomes (Khoza, 2013). The approaches are; teacher-centred, learner-centred or content-centred. These approaches should not be used in isolation. If the lecturer wants a suitable approach for presentation, then the teacher-centred approach would suffice. Alternatively, if the lecturer wants to grade content, then the content-centred approach would do. If the lecturer wants to contextualize learning, then the learner-centred approach would be adequate. However, the lecturer can incorporate all three approaches when teaching, for maximum effect.

When asked the following questions: Do you know that the NCV LO Computer Skills curriculum demands a learner-centred approach to teaching? Would you say that in your teaching you follow these demands? To what extent? The participants responded as follows:

Mrs Sithole: So teacher-centred I would say, I teach and then students complete their tasks. Specific approaches are incorporated by those who were trained to teach Computer Skills.

Mrs Mpisi: Teacher-centred, as the teacher is in charge of the lesson.

Ms Khan: As I teach a practical subject – Computer Skills, I first lecture, then demonstrate using the data-projector. So I would say I use the teacher-centred approach i.e. behaviourism. I use the telling method and drilling to some extent to make sure the learner is being taught.

Mr Guy: The curriculum is centred on the teacher. I am not familiar with the learner-centred approach to teaching, as I have always taught this way.

The data revealed that participants were not familiar with the different teaching approaches used in teaching students Computer Skills in NCV. The basis for non-familiarity with other approaches could be that participants are only familiar with the traditional teacher-centred method. The traditional teacher-centred method revolves around the teacher telling the students what to do. This is evident in Ms Khan's response when she indicated that she uses the 'telling method'. All participants indicated that they used the teacher-centred approach and did not indicate using the learner-centred approach, nor the content-centred approach. Mr Guy's response that 'I have always taught this way' shows that he has not familiarised himself with the requirements of NCV, which is based on a

learner-centred approach. As was indicated in the discussion of the participants' profiles, Mr Guy had been long teaching prior to the introduction of the NCV in the year 2007. It seemed that he had no intention of moving away from his comfort zone. Mrs Mpisi is the only lecturer who began teaching after the inception of NCV, but she too also stated that she follows a teacher-centred approach only.

Another point of concern regarding the content-centred approach was that none of the participants mentioned the approach. Weimer (2007) argues that understanding and knowing content and being able to teach it are two different things. This implies that, over and above their content knowledge, Computer Skills lecturers need to have approaches to effectively impart knowledge to the students i.e. pedagogical knowledge. Pedagogical knowledge (PK) incorporates the teaching approaches and practices of conveying the content knowledge to students. TPACK as the framework for this study suggests that pedagogical knowledge in the form of teaching approaches is paramount for successful teaching and learning. It is suggested by Anderson and Elloumi (2004) that all three approaches of teacher-centred, content-centred and learner-centred should be incorporated according to their advantages and used together for effective teaching. None of the approaches should be used in isolation but together they can enhance teaching. Figgs' (2009) case studies on lecturers' excellence in teaching, revealed that lecturers' excellence was achieved by using learner-centred approaches.

4.2.2 Theme Two: Availability and unavailability of Adequate Resources

A resource by definition is anything that facilitates learning. The resources that are used for the effective teaching of Life Orientation Computer Skills include data projectors, overhead projectors, chalkboard, whiteboard, textbooks, computers, computer software, internet, Microsoft Powerpoint, Microsoft Access, Microsoft Word and the internet (DoE, 2007). When asked about the experiences in teaching LO Computer Skills with regard to resources available, Ms Khan responded by saying: When it comes to the practical component, the resources are scarce and out-dated. The textbooks are not enough; students do not get enough exercises to practice on and have to share computers. With Computer Skills, you need to practice, practice and practice to develop the skill of typing, you cannot just theorise Computer Skills.

Ms Khan noted that resources were not enough and had to be shared amongst students. This would then suggest the resources are not readily available. Computers have to be shared by students. When probed further, Ms Khan also mentioned that the lack of resources during teaching did not allow students the opportunity to learn practical skills. This suggests that students were not active participants during practical lessons if they are sharing computers. Students become mere spectators and this is against the curriculum requirement which states that students must have access to all the necessary resources required (DoE, 2007). When students are involved in lessons, they take ownership of their own learning, unlike when the lecturer tells them what to do all the time and they remain passive (teacher-centred). Life Orientation Computer Skills is mainly practical in nature and the College has the task of supplying students with the resources as stated in the NCV curriculum. There is evidence in the data generated from one participant (Ms Khan) that the colleges' textbooks were not enough.

Mrs Mpisi indicated that the few resources that are available are often outdated, when she stated that

"We mainly use old software and hardware. The computers are old and students complain that they take a long time to boot up. The programmes we use are Ms Office 2007. This is out-dated as Ms Office 2010 is the latest version. This places students at a disadvantage as when they go out into the work place they are not familiar with the latest version of Ms Office 2010. This is the major problem with resources; the fact that they are old." Using out-dated computer programmes such as Microsoft Office as stated by Mrs Mpisi may result in lecturers imparting incorrect, non-relevant and out-dated content knowledge to students. Ms Khan did not explicitly identify the resources that she considered as out-dated. However, Mrs Mpisi identified not only the computers, but also the programmes as well to be old and out-dated. This would imply that the knowledge students obtained at College will also be outdated and this could result in students needing to be retrained when they go out to industry, at the expense of the employer.

Furthermore, Mr Guy mentioned that those available resources usually malfunctioned.

"The printer does not function adequately as it freezes when receiving too many instructions to print, as in examination sessions."

All printers should be in good working condition, as all students need to print their work in the computer rooms. Furthermore, the use of different textbooks as a result of not subscribing to the same publisher could present a challenge of availability of adequate resources. There is evidence in the data generated as stated by Mr Guy:

"Resources, such as textbooks do provide a challenge, as different Colleges use different textbooks e.g. some use Macmillan, some use Pearsons and other publishers."

According to the NCV curriculum, textbook publishers are non-specific and each college is at liberty to choose any publisher that they deem appropriate. The ideal situation would be for all colleges to use the same textbook as examiners set from a specific textbook when setting papers. Hence, if the college is using a textbook that is not used by the examiner, this places the students at a huge disadvantage when being assessed.

Computer Skills students have to work and do practical tasks/projects using their computer skills. To facilitate teaching, the lecturer requires a wide range of resources, since Computer Skills as a learning area, is practical in nature. According to the SAGs, it is the responsibility of the College to provide the student with the necessary tools to meet the National Certificate Vocational (NCV) demands (DOE, 2007). The participants shared the same view that resources had a pivotal role to play in the teaching of LO Computer Skills. Ornstein and Hunkins (2012) note that curricula can fail because of inadequate support in the form of materials and adequate equipment available. A similar case was stated by participants in this study. Jones and Moreland (2004) assert that the use of available adequate resources is one of the critical aspects that enhance the technological knowledge which is crucial for effective teaching and learning. A qualitative study conducted by Makole (2015) on implementation of the National Certificate Vocation Programme at Tshwane North FET College revealed lack of resources in the form of small classrooms, computer rooms having no internet access, under-resourced libraries and simulation rooms, shortage of lecturers and textbooks. Ali (2006) states that technical resources overhead such as projects, screens and anv modern device (computers/printers) are important and contribute to proper teaching and learning. Lecturers in this study indicated that the inadequate resources impacted adversely on teaching and learning in the classroom level.

4.2.3 Theme Three: Lecturer Training and Professional Development Support By Department of Higher Education and Training

Teaching Computer Skills requires the expertise of a competent educator in the computer field and one in possession of a computer qualification (DoE, 2013b, p. 13). The participants noted the following with regard to training:

We have received no formal training in Life Orientation, since it was introduced in 2007 (Mrs Sithole).

Lecturers need to receive training on an on-going basis in Computer Skills. How will our students fare in the workplace when we as lecturers are teaching them out-dated programmes? (Mrs Mpisi)

Training is essential as it would keep me up-to-date with the latest programmes used in industry. (Mrs Mpisi)

Training is essential in teaching and learning. (Mr Guy)

It seems that guidance by the Department of Higher Education with regards to training would positively contribute to how lecturers teach. The lecturers in this study felt that the training offered by the Department would be of a great help especially if the training was practically based as computers is a practical subject. Keeping abreast of changes in the field of Computer Skills is a necessity for lecturers since computer technologies are constantly changing. This is acknowledged in the 'up-to-date' training (Ms Khan) and 'on-going' training (Mrs Mpisi) as indicated by these two participants. The data showed that lecturers would benefit from on-going training that is work-based and up-to-date, as stated by Mrs Mpisi. 'Work-readiness' of students is the focus of Life Orientation (DoE, 2007). In all countries, teacher training and professional development is the cornerstone in teaching and learning (Jones et al., 2013).

To the question: In your opinion, does Department of Higher Education and Training provide any support for the training of lecturers in the teaching of the NCV in general or Computer Skills specifically? The following responses emerged:

Mrs Sithole: No. DHET officials only arrive at the end of the year to monitor the exam process. We have no communication with DHET with regards to training sessions. I do not know of any training, if any, that is provided by DHET.

Ms Khan: It would be nice if DHET provided training on the actual content of instruction delivery.

Mr Guy: During the course of the year when teaching and learning takes place, they are not visible at all.

Mrs Sithole revealed that DHET officials arrive at year-end. The pivotal question would be why they would arrive at this time of the college calendar? Could this mean that they value examinations more than the actual process of teaching and learning? Also are they merely concerned with the outcomes and learner performance? Lecturers, like Mrs Sithole would rightfully feel neglected by the same officials who only arrived at year-end. The lack of training was an area of concern for all the lecturers in this study. It appeared from the interviews that lecturers would be appreciative of any training provided by the Department of Higher Education officials, but such was non-existent. Instead, the lecturers, as indicated by Ms Khan, expressed their need for practical training workshops that would equip them with practical skills so that they could effectively deliver the curriculum at their college.

The findings suggested that the factors that negatively impacted or hindered the teaching of Computer Skills could be overcome if a concerted effort with regard to training by the Department officials was made throughout the year. Findings showed that DHET officials do not play a visible role with regard to LO Computer Skills instruction which concurred with a study conducted by Davis (2011), which found that the lack of support by administrators or managers was a hindrance to teaching and learning of subjects.

To the question: Are there support structures that have been implemented by the College in assisting lecturers with regard to Computer Skills? Probe - What kind of structures? the following responses emerged;

Mrs Mpisi: We have subject committee meetings with a subject head¹ who will try to assist lecturers in Computer Skills. Mr Guy: There are no support structures in place. The college mainly concentrates on the core subjects² of NCV and not on fundamental³ subjects which are not taken seriously at all.

¹ subject head-person in charge of subject

There are contradicting views in Mrs Mpisi and Mr Guy's comments about support structures. Whilst Mrs Mpisi's comment highlighted that lecturers supported one another, Mr Guy's comment seemed to be positioned on a blame mode, where the College gets blamed for lack of support. Mrs Mpisi stated that support is provided by a subject head who initiates meetings in order to offer assistance to lecturers. Mr Guy has stated that the subject was not given the importance that is needed to get the required assistance.

One of the reasons for the lack of support was evident when interviewing Ms Khan when she said:

*Filler*⁴ *subjects are not taken "serious" by management…* (Ms Khan)

Ms Khan noted that that filler subjects were not given the required level of importance and that was confirmed when management did not take the subject Life Orientation 'serious': Hence, it would then suggest that officials were not focusing on the 'serious' nature of Computer Skills. The information gathered by the study revealed how DHET officials did not acknowledge the 'serious' aspect of Life Orientation. The lack of responsibility by DHET officials regarding Life Orientation as important would then filter down to lecturers and ultimately students. According to a study conducted by Jacobs (2011), Life Orientation lecturers need to take responsibility for Life Orientation, by taking the subject seriously, thereby instilling in the students an appreciation for the subject. TPACK focuses on the combination of combining all three components of technology, pedagogy and content knowledge in teaching and learning. Hence, Life Orientation must be given a 'serious' platform by DHET officials to facilitate training and professional development that must be supported by the Department of Higher Education and Training.

² core subject-main subject to learning programme

³ fundamental subject-generic subject to learning programme

⁴ filler subject- a subject used to fill-up your timetable

4.2.4 Theme Four: Experiences of Teaching

While interviewing the participants it became clear how lecturers' experiences assist in teaching. The lecturers admitted that they relied on their experience to teach Computer Skills in the classroom. Lecturers' experiences are based on their content knowledge and rationale (reasons) for teaching as revealed by the study.

To the question: How would you describe your role as a lecturer teaching Computer Skills to students? (Rationale-teaching reasons), the below responses were received:

Mrs Sithole: I basically rely on my own experience and knowledge when teaching.

Mr Guy: I would like to think that I am contributing to my students' knowledge by using the practical work-based knowledge I obtained in my many years of experience in industry and teaching.

Mrs Sithole and Mr Guy relied on their teaching experience in teaching and learning of Computer Skills. Mrs Sithole relied on her 21 years teaching experience and Mr Guy on his experience from teaching in industry to facilitate teaching. Mrs Sithole was never trained to teach Computer Skills, but her 21 years teaching experience assisted her with teaching with regard to perseverance. Mr Guy's industry experience gave him the confidence in the teaching and learning process. The data has revealed that lecturers such as Mr Guy and Mrs Sithole had their understanding of Computer Skills based on their own experiences. However, the content knowledge of the subject needed to be further elaborated. The focus of TPACK requires the successful implementation of content knowledge (CK).

Ms Khan responded with the following regarding her content knowledge:

The most obvious digital technology is difficult for me, as I am from the old school and find it hard to learn new things (Ms Khan).

Mrs Sithole responded with the following:

I did not feel confident to teach Computer Skills as I felt I will not do justice...

Both Ms Khan and Mrs Sithole's responses indicated that they did not feel confident teaching the content of Computer Skills. These responses suggest that the participants who received no Life Orientation training were less keen to teach Life Orientation. Although Ms Khan had also not received any formal LO training, she at least covered the basics of computer practice in her ACE studies:

Many years ago during my ACE training, I received a basic computer skill certificate... (Ms Khan).

In addition, Mrs Mpisi was self-taught with regards to Computer Skills:

I have taught myself the Computer Skills component. (Mrs Mpisi)

The responses concurred with the existing knowledge proposed by Rooth (2005), who indicated that lecturers would feel bitter about having to teach Life Orientation if they did not have the adequate content knowledge. Therefore, specialists in Life Orientation were needed (Rooth, 2005; Van Deventer, 2007; Van Deventer & Van Meeker, 2008). The study further revealed that lecturers' experiences were based on their rationale (reasons) for teaching. As suggested by Berkvens et al. (2014), the rationale for teaching should consist of personal reasons, content reasons, and societal reasons. In this study, participants' responses only included personal and societal reasons. Not one of them responded with content reasons, as shown below:

Mrs Sithole: They allocate you a subject just to fill up your teaching load....

Mrs Mpisi: I am teaching so that my students are in line with industry and am able to use the correct software when they go out to the workplace.

Ms Khan: Many years ago during my ACE training, I received a basic computer skill certificate... this could be the only reason I can think of for the rationale for teaching...

Mr Guy: I worked in industry prior to coming to the college...maybe based on my work based experience...

Mrs Mpisi and Mr Guy both used societal reasons to explain their rationale for teaching. Both based their rationale for teaching on other sources like industry. Mrs Sithole, however, was aware that she was only given Computer Skills to fill up her time table. Ms Khan only assumed she was allocated to teach Computer Skills based on her basic Computer Skills certificate. Both Mrs Sithole and Ms Khan used personal reasons as their rationale for teaching. Personal reasons would encompass knowledge that is unique to each lecturer and hold personal significance (Schiro, 2013). Societal reasons are mostly based on general knowledge, people's own views and verbal discussions. The reason that participants responded to only personal and societal reasons could be that NCV is a competence based curriculum where skills and higher-order knowledge is assessed. Life Orientation in the NCV programme "is underpinned by the following principles: an outcomes-based approach to education; high knowledge and skills emphasis; integrated and applied competence" (DoE, 2007, p. 2). Rodrigues and McKay (2010), however, argue that teaching experience does not guarantee that the educator will be an expert in a particular learning area. While lecturers' experiences facilitated teaching and learning in this study, it was also apparent that these lecturers needed to develop extensive knowledge related to the nature of Computer Skills (content knowledge), technological knowledge and pedagogical knowledge (Jones, 2002).

4.2.5 Theme Five: Time Allocation for Teaching and Learning

The NCV curriculum stipulates two periods a week for the teaching of LO Computer Skills. In this study, the time stipulated for teaching the subject and learning outcomes is insufficient with no room for any deviations, in Life Orientation as per Subject Guidelines document. This is evident in the following response by Mrs Mpisi:

Another challenge is the time allocation for teaching Computer Skills. We only have two periods a week for teaching Computer Skills, which is never enough time. In two periods, by the time the student sits down and puts on the computer and most computers are slow, about a quarter of the time is gone.

Computer Skills is practical and students need more than two periods a week to grasp the skill. It would seem that resources like hardware and software that are old, place additional time constraints in the way of logging on and starting up, to the already limited time available for teaching and learning. Also indicated by Ms Khan and Mr Guy:

Ms Khan: We are all so busy conducting assessments. After the assessments, we have to mark and this takes up so much of our time. As you have just completed one assessment and marked it, the next assessment is scheduled to be done.

Mr Guy: A lot of time is spent on setting assessments...

When probed further, Mrs Mpisi also confirmed that assessments consume much of teaching time.

There are seven assessments for the year in fundamental subjects like Life Orientation.

The data revealed that lecturers spend a lot of time on other activities, such as conducting assessments, and the time they have for teaching is reduced. These activities would also include marking and setting of the assessment tasks as stated by Ms Khan, Mrs Mpisi and Mr Guy. Participants indicated that the time allocated for teaching was insufficient.

When Mr Guy was further probed on whether there were any challenges that he experienced in the teaching of Level 2 Computer Skills and if he could please elaborate on how he dealt with these challenges, the following was his response.

Furthermore, the students do not have a background knowledge of computers which makes it very difficult for them to understand and flow in the subject. It would be better if students came with a fundamental knowledge of Computer Skills, so as to save time for teaching.

From the data generated, it would appear that students, who do not have a basic knowledge of computers, would consume much of the lecturers' time when teaching. Lecturers have to spend more time explaining and demonstrating basic computer literacy skills. Mr Guy suggested that the basic Computer Skills knowledge by students would be a help in saving teaching time. Mrs Mpisi shared the same view that other matters that are College related consume much of the time allocated for teaching and learning. She claimed:

We have subject committee meetings with a subject head who will try to assist lecturers' in Computer Skills. The subject meetings are scheduled for once a month. Other activities on the college programme include; sports, exam sessions, assessment week. These are all time consuming. The data revealed that non-academic activities such as meetings and sports events take up much time on the College programme, as stated by Mr Guy and Mrs Mpisi. Studies have revealed that, "such strict guidance to times stipulated for teaching and learning, tends to raise learning outcomes in the short term, but demotivates teachers and does not allow for individualised learning" (Berkvens & Van den Akker, 2014, p 18). Berkvens (2014) further states that, "formal time for learning is often still spent at school. Timetables are rather conservative, while many opportunities are ignored" (p. 18). The participants in this study indicated that the Subject Guidelines document allocated too little time for lecturers of Computer Skills to complete the content of the NCV curriculum. Furthermore, having to conduct seven assessments (formal and informal) for the year was time-consuming, so the time especially for the Computer Skills component is way too short. The pedagogical aspect of TPACK is unfavourably impacted as the lecturers do not have adequate time to implement the technological and content knowledge which are inseparable.

4.2.6 Theme Six: Lecturer Communication and Collaboration

According to the Subject Guidelines, Computer Skills can be successfully taught if all those involved in lecturing are able to partner with other colleagues as part of a team (DoE, 2007). For this reason, more can be achieved through team interaction and team work amongst the Life Orientation team rather than each lecturer working in isolation.

When participants were asked the question: Do experienced/expert Computer Skills lecturers assist their fellow lecturers in the teaching of Computer Skills?, they responded thus,

Mrs Sithole responded: Yes, every one helps where ever they can. Sometimes it is difficult to leave the computer venue and go and find another lecturer to ask to assist you. . .

Mrs Mpisi response: Yes. There is help from other lecturers if you need it. You can ask your colleagues for help with the content that you are not familiar with.

Ms Khan answered: Yes. I get assistance from whoever will assist me...'

Mr Guy indicated: Yes. I believe in 'ubuntu', mentoring of one another. This is used at our college, as my colleagues are always ready to help me if need be. In our African culture, we believe in lifting each other up and helping our community in assisting all who are in need of help. This principle is also applied at our place of work, where we see if someone is in need of assistance, we are committed to give support wherever it is needed.

It was evident in interviews that lecturers were also open to working with other lecturers. This was revealed when Mrs Mpisi indicated that there was help available if needed. Lecturers said that they wanted to work jointly with others to improve teaching and learning. The data generated suggested that participants were open to assist each other by communicating. During the interviews the participants revealed that they were also open to communication with other lecturers, as noted by Ms Khan who says she gets assistance from whoever is willing to assist her. Hall and Hord (1987) suggest that all must be interested in working together to improve teaching and benefit learners. Participants adhered to the Subject Guidelines by working together, to facilitate the teaching of the curriculum as they could share ideas, effective teaching methods and therefore benefit from each other. In the chosen framework of this study (TPACK), technological, pedagogical and content knowledge are inseparable. Lecturers benefited from each other by using team work as indicated by Mr Guys' response about using his colleagues to assist him by grouping students together indicates team work. Team work helps lecturers who are struggling with Computer Skills to cope with the subject content knowledge (CK in TPACK) as indicated by Mr Guy. Lecturers who are knowledgeable

about the subject content can teach the struggling lecturers' students together with his/her students.

Students are grouped together and the more knowledgeable lecturer of the subject content will then assist and teach the students. (Mr Guy)

Mr Guy indicated that the knowledge of subject content would determine the lecturers who are more 'knowledgeable'. Mrs Mpisi also indicated that communication was essential amongst lecturers, when she said; *We have subject committee meetings with a subject head who will try to assist lecturers in Computer Skills.* This agrees with Ornstein and Hunkins' (2012) suggestion that communication among colleagues needs to be promoted for successful teaching. All four participants expressed the need for collaboration during the individual interviews. The learning environments must also provide collaborative opportunities that support teaching and learning (Amory, 2010). This view is supported by Pillay (2012) who states that nobody should claim to have all the answers but lecturers should depend on the knowledge and insight of their colleagues in order to best support students.

4.2.7 Theme Seven: Computer Skills as a Vocational Subject

It is clear that the participants wanted to give their students hands-on skills in a specific field. Participants viewed Computer Skills as a vocational subject as they all placed emphasis on preparing students for a specific skill.

Mrs Sithole and two other participants had similar views when interviewed about preparing students for Computer Skills:

Mrs Sithole: My role is to teach a practical skill to students in LO L2. I am responsible for getting students to learn Ms Word, Ms Excel and Ms PowerPoint. Hence, I prepare students for life in the technological world. Students will also be taught how to be able to use the internet, and understand the role that it plays, find relevant information, process it, make decisions and learn how to use computers responsibly. A high level of competence is expected of students.

Mrs Mpisi: I am teaching so that my students are in line with industry and am able to use the correct software when they go out into the workplace. I am a practical person; hence I enjoy the practical component of Computer Skills. Computer Skills are essential to students as they help students with the basic issues of computer literacy.

Mrs Sithole and Mrs Mpisi also expressed their views on the fact that students need to acquire basic skills in the form of Computer Skills. In the data generated, Mrs Sithole wants to find better ways to reach and teach the students like giving them vocational skills like Ms Word, Ms Excel and Ms PowerPoint. Furthermore, Mrs Sithole stressed the importance of acquiring extra skills like knowledge of the internet as a search engine which will assist students immensely in researching information. She says students will benefit from the knowledge of the internet as they can research information for their college projects. Mrs Mpisi expressed the view that she was teaching so that students would be familiar with the correct programmes when they entered the workplace. She also said that Computer Skills were a basic vocational skill that students needed when they went out into industry. Ms Khan also expressed the need for skills training to assist students in the work place as indicated by her response to the following question: How would you describe your role as a lecturer teaching Computer Skills to students? (Rationale-teaching reasons):

Hence, the rationale for teaching Computer Skills is that it is a very practical skills-based subject which can help students in the work place.

Ms Khan indicated that by her teaching Computer Skills, she would assist students to obtain skills in computers which would help in the work place. Her emphasis was on the practical nature of Computer Skills. Vocational skills will give more students a chance for employability as students will have a hands-on skill. Jones et al. (2013) assert that in some countries, the vocational emphasis is politically driven, like when there is a strong demand for skilled labour. Likewise, it is important to note that in the South African context, regarding TVET colleges, vocational emphasis is also politically driven. The New South Africa, after 1994, prompted the new curriculum programme of National Certificate Vocational (NCV) to be implemented in 2007. The new curriculum of NCV was implemented to address the shortage of skills in the labour market and industry. This is in accordance with the department statement, "A view to strengthening it as the main route for general vocational education. The Department of Higher Education and Training is committed to the NCV" (DHET, 2012, p. 22).

4.2.8 Theme Eight: Subject Specialisation

The issue of Subject specialisations came up when lecturers were interviewed. There is evidence from the data that educator subject specialisation has an impact on the teaching of Computer Skills at college.

To the question: How would you describe your role as a lecturer teaching Computer Skills to students? (Rationale-teaching reasons), Ms Khan responded with the following:

Many years ago during my ACE training, I received a basic computer skill certificate, this could be the only reason I can think of for the rationale for teaching. I have also taught CAT at a high school. At schools the subject that is linked to Computer Skills is Computer Applications Technology (CAT). However, this is not a designated subject, meaning it cannot be used to gain access to university. Students also complete the ICDL (International Computers Drivers Licence) course during Grade 10 and 11 at basic education level. Hence, the rationale for teaching Computer Skills is that it is a very practical skills-based subject which can help students in the work place. Ms Khan noted that her ACE training has assisted her in teaching Computer Skills. *My Adult Certificate in Education (ACE) has greatly helped me in teaching Computer Skills. I was taught the basics of computer literacy when I enrolled for this certificate at Unisa.* Going further, Ms Khans' experience, obtained in the subject CAT from her school teaching, assisted her in specialising in Computer Skills. The data suggests that a lecturers' subject specialisation assists them in understanding some concepts in Computer Skills. The issue of the subject specialisations came up even when lecturers were asked questions unrelated to it. When Mrs Sithole was asked: Are there any challenges that you experience in the teaching of Level 2 Computer Skills? Please elaborate. How do you deal with these challenges? She responded;

We have received no formal training in Life Orientation, since it was introduced in 2007, they allocate you a subject just to fill up your load. I did not feel confident to teach Computer Skills as I felt I will not do justice as the subject requires specialised knowledge which requires specialised training which I do not have.

Mrs Sitholes' response indicates that she sees the need for specialisation, which would help her feel confident to teach Computer Skills. Mrs Sitholes' response was similar to Ms Khans' response in that another subject, in Ms Khans' case CAT influenced how she taught Computer Skills to her students. The data generated suggests, as indicated by Mrs Sithole, that the subject specialisation was beneficial as it assisted them to comprehend some Computer Skills content and be in a position to deliver this content in a Computer Skills classroom. Subject specialisation is imperative in the NCV curriculum as SAGs suggest that "Computer Skills lecturer must be computer literate and have an advanced knowledge of the following programmes: Word processing, Spreadsheets, Presentation and knowledge of the Internet and e-mail" (DoE, 2013b, p. 13). The focus of TPACK is on, amongst other aspects, the successful implementation of the content knowledge.

4.2.9 Theme Nine: Status of Computer Skills in comparison to gateway subjects in NCV

The status of the Computer Skills as a subject in the college seems to concern lecturers in various ways. There is evidence in the data generated to support the status of Computer Skills as indicated by the following responses:

Mrs Mpisi said: Computer Skills are essential to students as it helps students with the basic issues of computer literacy. These basic Computer Skills should start early at school level so by the time they come to College they familiar with the relevant technology. (Mrs Mpisi)

Mr Guy: Furthermore, the students do not have a background knowledge of computers which makes it very difficult for them to understand and flow in the subject.

Mrs Mpisi and Mr Guy recommended that students receive basic training in Hence, Mrs Mpisi recommends basic Computer Skills as a schools. prerequisite for entry into TVET colleges. From the data generated, it would appear that students who have not had any basic knowledge of computers, would consume much of the lecturers' effort when teaching. At TVET Colleges, the policy is that students who have completed grade nine can gain entry into the NCV programme (DoE, 2007). Hence, students might not have had any basic Computer Skills training when they enrol at TVET Colleges. Furthermore, learners who are not successful at school are all encouraged by the Department of Education to go to TVET Colleges. This poses a concern in that if students did not pass Grade 10, 11 or 12 it is assumed that they are not academic and are informed to go to TVET Colleges so as to do skills training. Hence, TVET Colleges' have an assortment of students ranging from grade 10 upwards. This would imply that the students who come to TVET Colleges without the basic Computer Skills knowledge affects the status of the subject Computer Skills. Siksin (2001) contends that school subjects like Maths, English, and Science are regarded as basic in the curriculum. These high status subjects may receive

more resources and power within the school and community than lower status subjects like Computer Applications Technology (CAT). The same argument is applicable at TVET level, as expressed by Mr Guy remark:

Likewise with basic education, life orientation is marginalised and not given preferential treatment like subject of maths and science at TVET College.

The participant, Mr Guy, revealed that the low status had far-reaching effects for their college. The Management at the college did not prioritise Computer Skills in giving the subject preferential status. Lecturers are negatively affected by the lower status of Computer Skills. Participants expressed that this could be the result of stakeholders who are not taking Computer Skills seriously at TVET level. It has filtered down to colleges as a result of Life Orientation not having the grading status to obtain points at basic education level to progress to tertiary education, as indicated by Ms Khan:

Ms Khan: At schools, the subject that is linked to Computer Skills is Computer Applications Technology (CAT). However this is not a designated subject, meaning it cannot be used to gain access to university. Students also complete the ICDL (International Computers Drivers Licence) course during Grade 10 and 11 at basic education level.

Life Orientation in the FET phase at schools does not allow learners to gain entry into University, as observed by Ms Khan. Ms Khan suggested that the status of Computer Skills is affected by the non-progression into universities. Hence, more should be done to get Computer Skills the necessary grading. Researchers like Christiaans (2006) and Rooth (2005) agree that knowledge and skills of the students determine the status of the subject. De Klerk and Van der Walt (2006) concur with the above mentioned researchers that the status of the learning area/subject depends on the priority placed on the subject.

4.3 CONCLUSION

This chapter outlines the analysis of the data generated from participants in the form of semi-structured interviews. These were analysed using the literature reviewed and the theoretical framework. This chapter also presented and analysed findings by focusing on the themes that were generated from data. Use was made of thematic analysis during the analysis of data, and numerous themes were found which are discussed in detail in the chapter. These themes were: familiarity and non-familiarity of teaching approaches, availability and unavailability of resources, lecturer training and professional development supported by Department of Higher Education and Training, lecturer experiences, time allocation for teaching and learning, lecturer communication and collaboration, Computer Skills as a vocational subject, subject specialisation and status of Computer Skills in comparison to gateway subjects in NCV. Chapter five provides a summary, conclusion and recommendations that have been made from the findings of the study.

CHAPTER FIVE

SUMMARY, FINDINGS AND RECOMMENDATIONS

5.1 INTRODUCTION

In chapter 4, the data is presented and analysed. In this chapter, a summary of the study, the findings and recommendations will be discussed. In view of this study, focus is on the lecturers' experiences of teaching Computer Skills at a TVET College in a township. The study was situated in the province of KwaZulu-Natal. The research questions were as follows:

5.1.1. What are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

5.1.2. Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

5.1.3. How do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching?

5.2 SUMMARY OF THE STUDY

Chapter one provided an outline of the study. This was done by highlighting the many post-1994 changes that occurred in the political, educational and social landscape of South Africa. With these educational changes came many changes taking place affecting teaching experiences of lecturers in South Africa. The National Department of Education experienced pressure in providing a curriculum that would address the past imbalances in education. A background, research problem and purpose of the study were provided.

Chapter two presented the literature review on the study. The theoretical framework of TPACK was also discussed. TPACK was the most suitable framework for this study as it provided the components of knowledge which are specific to lecturers' experiences in teaching Computer Skills. It became apparent that lecturers who teach Life Orientation were not trained to teach the subject but use their knowledge and experience, that is, the ideological-ware

(IW) to teach the subject. It was discovered that the software (SW) resources, particularly the Internet, are enjoyed by most lecturers and students and are being used pedagogically to promote teaching and learning.

Chapter three gave the research design and methods used. Chapter four outlined the analysis and interpretation of data that was generated through interviews.

5.3 FINDINGS

In this section, I elaborate how the findings connect to the three critical questions of this study. Two of the questions are addressed together, namely; what are lecturers' experiences of teaching Level 2 LO Computer Skills in the NCV curriculum and how do lecturers' experiences of teaching Level 2 LO Computer Skills in NCV impact their teaching? These findings show a compelling picture of lecturers' experiences in teaching and how these experiences impact their teaching.

As addressed in chapter four, lecturers encountered numerous hindrances when teaching LO Computer Skills. One of the hindrances was the lack of resources. Ornstein and Hunkins (2012) note that curricula can fail because of inadequate support in the form of materials and adequate equipment available. The findings revealed that because of the lower status of Computer Skills at the College where the research was done, the subject did not get the necessary resources needed to teach the subject effectively. De Klerk and Van der Walt (2006) concur that the status of the learning area/subject depends on the priority placed on the subject. Participants indicated that Computer Skills was regarded as a filler subject at the College and given to any lecturer who was short of teaching time on their timetable to meet the minimum teaching hours required. The College did not prioritise LO Computer Skills and failed to allocate adequate funds for effective teaching. Through the interviews, it emerged that lecturers were handling a subject that was practical in nature with very few necessary resources to support the teaching and learning processes. Jones and Moreland (2004) assert that the use of available adequate resources is one

of the critical aspects that enhance the technological knowledge which is crucial for effective teaching and learning. Over-enrolment of students resulted in students sharing computers. Furthermore, not all students would get the required practice needed for the Computer Skills component as a result of the shortage of computers. The available resources were also out-dated according to participants. TVET education focuses on an authentic learning experience, the use of out-dated resources does not adhere to the requirements of the NCV curriculum (DoE, 2013b).

The progression of LO from TVET College level into tertiary level was not clear for the lecturers. All LO components are insufficient for students to gain entry at a University as indicated by the findings of this study. The lack of progression undermines LO as a subject. The lack of career progression of LO was also highlighted by one of the participants who noted that as a result of Life Orientation not being weighted at a basic education level, students cannot gain entry into universities. Furthermore, the progression challenge had an influence on how Life Orientation was perceived at Colleges.

Participants agreed that the time stipulated for teaching the subject is insufficient with no room for any deviations, in Life Orientation as per Subject Guidelines document. It is suggested that, "such strict guidance tends to raise learning outcomes in the short term, but demotivates teachers and does not allow for individualised learning" (Berkvens & Van den Akker, 2014, p 18). The Subject Guidelines document allocates very little time for lecturers to complete the content of Computer Skills in the NCV curriculum.

The findings on the technological knowledge of lecturers suggest that there is a paucity of technological knowledge on the part of lecturers. The main contributing factor to the above is their lack of knowledge of teaching approaches. Lecturers are unable to implement the three main approaches in their teaching be it teacher-centred, learner-centred or content centred. None of the teaching approaches should be used in isolation but each should be used according to their strength. Knowing whether the curriculum is steered by the teacher-centred, content-centred or learner-centred approach enhances the

opportunity of achieving the attained curriculum due to the positive correlation between the implemented and intended curriculum (Hoadley & Jansen, 2014). Hence, due to lecturers' scarcity of knowledge in teaching approaches, students of Computer Skills are technologically disadvantaged in their learning. Similar disadvantages are apparent in the findings of pedagogical knowledge (PK). Weimer (2007) argues that understanding and knowing content and being able to teach it are two different things. Khoza (2001) suggests the need for linking learning outcomes to assessment methods.

Why do lecturers have these experiences of teaching Level 2 LO Computer Skills in the NCV curriculum?

The findings on lecturers' reasons for teaching suggest that lecturers were not knowledgeable on whether their reasons for teaching were personal, content or societal. Personal reasons are when lecturers teach to incorporate environments that assist students in learning (Khoza, 2015b). Students are then able to create their own unique knowledge that is personal to them and has private significance to each of them (Schiro, 2013). Hence, knowledge is seen as the basis of learning which is not separate from the student. Societal reasons are influenced by personal reasons. Societal reasons are a result of attaining noticeable and measurable results which is the essence of the NCV curriculum.

Other findings also suggest that for the participants in this study, teaching experience and self-taught skills can be used to teach students effectively and promote learning. Teaching should incorporate authentic teaching resources, as teaching is not about resources where students learn from resources, but it is about ideology where they learn with resources (Amory, 2010).

Inadequate support by authorities was also a concern expressed by participants. Participants indicated that management did not prioritise LO Computer Skills, hence they did not receive the necessary support. This was evident in the lack of resources available when teaching. In order to enhance teaching and learning in Computer Skills at Colleges, lecturers need to be supported. The lack of teacher training and professional development was also

evident as a result of lack of support by DHET. Lecturers indicated that they would immensely appreciate hands-on training, as Computer Skills are a practical subject that required many practical skills. They lacked the practical skills and were unable to assist students in the learning area. Insufficient training and support for lecturers would lead to them becoming frustrated with teaching and learning (Fullan, 1992).

All participants expressed the need for collaboration during the individual interviews. This would meant that Computer Skills can be effectively taught if all those involved in lecturing are able to collaborate and communicate with each other. Pillay (2012) states that nobody should claim to have all the answers but educators should depend on the knowledge and insight of their colleagues in order to best support students. The learning environments must also provide collaborative opportunities that support teaching and learning (Amory, 2010).

The findings on content knowledge showed that lecturers had not acquired adequate content knowledge. This is evident in lecturers not being able to implement or even differentiate most teaching approaches. The NCV curriculum is mainly concerned with local and verbal views of everyday conversations. Alternatively, the Nated curriculum is performance based and tests that which is cognitively absent in a student. In the Nated curriculum, students are expected to learn from the bottom order knowledge to the top order knowledge. The problem is exacerbated when lecturers have to teach across both NCV and Nated curricular simultaneously.

Another essential finding was that the lecturers were allocated Life Orientation as a 'filler' subject in order for them to meet the minimum number of lecturing hours. The allocation of 'filler' subjects has a serious impact on the lecturer concerned and their teaching since the lecturers are usually not passionate about the subject and appear to be more disinterested than their colleagues.

It will always be a challenge for TVET college lecturers to assist learners in Computer Skills if they do not have technological, pedagogical and content
knowledge (TPACK), as both the NCV and the NATED curriculum employ these three types of knowledge (Khoza, 2015c).

5.4 RECOMMENDATIONS

On the basis of the study findings, the study proffers the following recommendations:

As the findings indicate that students are technologically disadvantaged due to lecturers' dearth of knowledge concerning teaching approaches, it is recommended that lecturers form teams where they can assist each other and contribute to the 'best practice' of teaching Computer Skills. Lecturers should also meet often at subject meetings so as to brainstorm on how to implement the three approaches: teacher-centred, learner-centred and content-centred. The use of team work employing ideas, suggestions and practical ways on teaching across these three approaches can be utilised. As a result of the lecturer gaining practical experience, they will be able to employ the importance of teaching and learning and they will work towards implementing the learner-centred approach. Students will be active participants in the lesson with the lecturer just acting as a 'facilitator' to promote learning. Knowledge of the three approaches will increase the link between the implemented, intended and students achieving the attained curriculum (Hoadley & Jansen, 2014).

It is also recommended that lecturers plan ahead for teaching. Planning ahead will reduce the dilemma of teaching the two curricula simultaneously and also ensure that learning focuses on the student. Keeping abreast on the latest hardware and software versions is pivotal for lecturers to ensuring that students get relevant hands-on training that they can apply in the workplace.

The Department of Higher Education and Training should embark on frequent training workshops for lecturers in the TVET sector to provide the needed support and mentorship.

Lecturers should be given adequate teaching and learning resources by the College to ensure that students are prepared for learning.

Teaching should become more learner-centred, as the NCV curriculum is more learner-centred in nature. The intention is to transform curricula to meet the needs of commerce and industry. Lecturers who revert to the lecture method undermine the purpose of the NCV curriculum which is learner-centred and are not preparing students for the workplace environment.

The College needs to invest in a heavy duty printer to cope with the overload of printing during examinations.

Furthermore, TVET Colleges must also take responsibility for developing lecturers' knowledge and skills in the learning area of Computer Skills by hosting workshops.

It can be suggested that all non-teaching activities could be scheduled after College hours so as not to deplete on teaching and learning time.

5.5 IMPLICATIONS FOR FUTURE RESEARCH

The study was small in nature focusing on lecturers' experiences at a TVET College. Only four lecturers at the College were used for this study. Like other qualitative studies, it cannot be generalised to other contexts. However, this did not affect the quality of the findings. I suggest that in future, a similar study can be conducted on more than one TVET colleges. Further research should lead to a more pro-active approach in developing better teaching practices for TVET colleges based on lecturers' experiences.

5.6 LESSONS LEARNT FROM THIS STUDY

I had mixed feelings about approaching lecturers to participate in this study as they are my colleagues. I was surprised by them agreeing to take part in this research. I learnt that this could be the result of me informing them of the confidentiality part of the study and that they would not be targeted based on the views they expressed. By communicating with fellow colleagues, I was cognisant of the fact, that this was not personal, it was just the nature of research.

My lack of knowledge on interviewing skills was tested and I now feel much more confident with regard to this research method. Conducting semi-structured interviews proved to be much longer and tedious than I first thought it would be. Also, transcribing of data collected from participants when interviewed was a never ending task for me. Hence, I learnt patience which I did not know I possessed and perseverance essential in conducting research in general. Again, seeking consent from the relevant parties to conduct my research also proved to be more difficult than I anticipated. I had to exercise much patience until I was eventually able to proceed.

Conducting this study allowed me the opportunity to become more informed about my topic; lecturers' experiences in teaching of LO Computer Skills. Understanding different experiences from lecturers was very rewarding, as it gave me insight as to how they viewed teaching Computer Skills. My knowledge of Computer Skills was vastly improved due to the numerous articles and scholarly works that I had to engage with for this study. It all proved to be very rewarding and informative.

5.7 CONCLUSION

In this study, I sought to explore lecturers' experiences of teaching Life Orientation at a TVET College. The study revealed lecturers' unfamiliarity with teaching approaches; unavailability of resources; lecturer content gaps; insufficient time to teach content; lack of continuation of Life Orientation to tertiary education; and the lower status of Computer Skills compared to other subjects. The following findings revealed why lecturers had the experiences they do: lack of training and professional development; lack of support at College level; and the need for communication and collaboration. The study also revealed how lecturers' experiences impacted their teaching of LO Computer Skills. The main thing that impacted their teaching experience was their lack of knowledge on teaching approaches. Lecturers were unable to implement the three approaches in their teaching. Students were negatively impacted in learning with regards to lecturers' paucity of knowledge of teaching approaches. Understanding these experiences can help the College to assist lecturers in teaching and learning. Although lecturers were doing the best they possibly can, they were overwhelmed by a large number of negative factors that impacted their teaching experiences. Also, the best teaching approaches for lecturers were showcased in this study to help lecturers with the teaching of LO Computer Skills in the NCV curriculum.

REFERENCES:

- Adam, S. (2004). Using learning outcomes: A consideration of the nature, role, application and implications for European education of employing learning outcomes at the local, national and international levels. Report on United Kingdom Bologna Seminar, July 2004, Herriot-Watt.
- Anderson, T., & Elloumi, F. (2004). *Theory and practice of online learning*. Canada: Athabasca University.
- Amiel, T., & Reeves, T. C. (2008). Design-based research and educational technology: Rethinking technology and the research agenda. *Educational Technology & Society, 4*(11), 29-40.
- Amory, A. (2010). Education technology and hidden ideological contradictions. *Educational Technology & Society, 13*(1), 69-79.
- Altrichter, H. (2005). Curriculum implementation limiting and facilitating factors. *Context Based Learning of Science*, 33(3), 35-62.
- Ball, D. L., & McDiarmid, G. W. (1990). The subject matter preparation of teachers. In W. R. Houston (Ed.), *Handbook of research on teacher education* (pp. 437-449). New York: Macmillan.
- Bauer, N. W., & Gaskell, G. (2003). *Qualitative researching with text image and sound.* London: SAGE.
- Bazeley, P. (2009). Analysing qualitative data: More than 'identifying themes'. Retrieved from www.researchsupport.com.au/More than themes. pdf.
- Bernstein, B. (1999). Vertical and horizontal discourse: An essay. *British Journal of Sociology of Education, 20*(2), 157-173.

- Bondy, A. (2007). The intended and interpreted technology curriculum in four New Zealand secondary schools: Does this all mean the same? A thesis submitted in partial fulfilment of the requirements for the Doctoral dissertation, University of Massey.
- Bonvin, J., & Galster, D. (2010). Making them employable or capable? Social integration policies at the crossroads. Farmington Hills: Barbara Budrich Publishers.
- Brown, C. (2007). Disciplinary differences in the use of educational technology.In L. Czerniewicz, *International Conference of E-Learning*. New York: Routledge.
- Brown, C. (2008). Trends in student use of ICTs in higher Education. In L. Czerniewicz, South Africa 10th Annual Conference of WWW Applications. CapeTown: South Africa.
- Brown, C. (2008). The impact of South Africa's ICT infrastructure on higher education. In H. Thomas, A. van der Merwe, & L. van Dyk, *International Conference of E-Learning.* Cape Town: South Africa.
- Carl, A. E. (2009). *Teacher empowerment through curriculum development: Theory into practice*. Cape Town: Juta and Company Ltd.
- Christians, C.G. (2010). Ethics and politics in qualitative research. In N.K. Denzin & Y.S. Lincoln (Eds.), *Handbook of Qualitative Research* (pp. 133–155). Thousand Oaks, CA: SAGE.
- Christiaans, D. J. (2010). Empowering teachers to implement the Life Orientation Learning area in the Senior Primary phase of the General Education and Training Band. Unpublished Masters dissertation, University of Stellenbosch.

- Cohen, L., Manion, L., & Morrison, K. (2011). *Research methods in education* (7th Ed.). New York: Routledge.
- Coombs, P. H. (1985). *The world crisis in education: The view from the eighties*. New York: Oxford University Press.
- Conrad, C. F., & Serlin, R. C. (2011). The Sage handbook for research in education: Pursuing ideas as the keystone of exemplary inquiry. London: SAGE.
- Crabtree B. (2006). *Qualitative Research Guidelines Project*. Retrieved from <u>http://www.quaires.org/HomeoBSE-3594.html</u>
- Creswell, J. W. (1994). Research design: Qualitative and quantitative approaches. London: SAGE.
- Creswell, J. W. (2012). Qualitative inquiry and research design: Choosing among five approaches. California: SAGE.
- Creswell, J.W. (2014). Qualitative inquiry and research design: Choosing among five approaches. In I. Christian, C. Bertram, & S. Land (Eds.), *Understanding research: An introduction to reading research (pp.1-45)* California: SAGE.
- Czerniewicz, L., K. Williams & C. Brown (2009). Students make a plan: Understanding student agency in constraining conditions. *ALT-J. The Association for Learning Technology Journal, 17*(2), 75-88.
- Davis, E. (2011). Teachers' experiences in the implementation of Technology Education curriculum in one secondary school in the St George East District in Trinidad. Masters dissertation, University of the West Indies.

- Dawson C. (2006). A practical guide to research methods: A user friendly manual for mastering research techniques and projects (2nd ed.). New York: Oxford University Press
- Day, T., Raven, M., & Newman, M. (1998). The effects of worldwide web instruction and traditional instruction and learning styles on achievement and changes in student attitudes in a technical writing in agricommunication course. *Journal of Agricultural Education*, 39(4), 65-75.
- Denscombe, M. (1998). *The good research guide for small-scale social research projects*. Buckingham: Open University Press.
- Denzin, N & Lincoln, Y. (2003). Introduction: the discipline and practice of qualitative research. In N. Denzin & Y. Lincoln (Eds.), *The landscape of qualitative research: Theories and issues* (pp.1-45). Thousand Oakes, CA: SAGE.
- Dewey, J. (1933). *How we think: A restatement of reflective thinking to the educative process.* Boston, MA: Heath.
- DHET. (2012). *Green paper for post-school education and training*. Pretoria: Government Printing Works.
- Dicicco-Bloom, B., & Crabtree, B. F. (2006). The qualitative research Interview. *Medical Education*, *40*(4), 314-321.
- Department of Education. (2007). Higher Education: National Certificate (Vocational): Subject Guidelines; Life Orientation, NQF Level 3. Pretoria: Government Printing Works.
- DOE. (2013b). Subject Guidelines. Pretoria: Government Printing Works.

Donnelly, R., & M. Fitzmaurice. (2005). Designing modules for learning. In G. O'Neill, *Emerging issues in the practice of university learning and teaching*, (Eds.), (pp.241-247). Dublin: AISHE.

- ECDL Foundation (2011). *Objectives of the benefits of the ECDL*. Retrieved from http://www.ecdl.com/syllabus/index.html
- Figgis, J. (2009). Regenerating the Australian Landscape of Professional VET Practice: Practitioner-driven changes to teaching and learning. A National Vocational Education and Training Research and Evaluation Programme Report. Adelaide: ERIC.
- Finch, C. R., & Crunkilton, J. R. (1999). Curriculum development in vocational and technical education planning, content, and implementation. Needham Heights: Allyn and Bacon.
- Fortune, A. E., Reid, W. J. & Miller, R. (Eds.). (2013). *Qualitative research in social work*. London and Chichester, UK: Columbia University Press.
- Gamble, J. (2006). *Knowledge, curriculum and qualifications for South African Further Education.* Pretoria: Human Sciences Research Council (HSRC) Press.
- Gray, D. E. (2004). Doing research in the real world. London: SAGE.
- Goldney, D., Murphy, T., Fien, J., & Kent, J. (2007). Finding the common ground: Is there a place for sustainability education in VET? *A National Vocational Education and Training Research and Evaluation Programme Report*. Adelaide: NCVER.
- Grollman, P. & Rauner, F. (2007). Preface. In P. Grollmann, & F. Rauner, International perspectives on teachers and lecturers in Technical and Vocational Education (pp. xi-xii). Dordrecht: Springer.

- Guba, E. G., & Lincoln, Y. S. (1994). Competing paradigms in qualitative research. In N. K. Denzin & Y.S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 105-117). Thousand Oaks, CA: SAGE.
- Guest, G., MacQueen, K. M., & Namey, E. E. (2011). *Applied thematic analysis*. Los Angeles: SAGE.
- Hardman, J. (2008). Researching pedagogy: An activity theory approach. *Journal of Education, 45*(2), 65-95.
- Henning, E. (2004). *Finding your way in qualitative research.* Pretoria: Van Schaik.
- Hough, J. & Neuland, E. (2014). Strategic management and the use of Information and Communication Technologies by selected South African and American students. South African Journal of Higher Education, 28(2), 455-465.
- Hughes, J. (2005). The role of teacher knowledge and learning experiences in forming technology-integrated pedagogy. *Journal of Technology and Teacher Education*, *13*(2), 277-302.
- Hunkins, F., & Ornstein, A. C. (2012). *Curriculum foundations, principles, and theory.* Boston, MA: Pearson.
- Ismail, S. A. (2014). Factors affecting the implementation of information literacy education in Malaysian primary schools. New Zealand: Victoria University of Wellington.
- Jephcote, M., & Salisbury, J. (2009). Further education teachers' accounts of their professional identities. *Teacher and Teacher Education*, 25(3), 966-972.

- Johnson, R. B., & Onwuegbuzie, A. J. (2012). Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, 33(7), 14-26.
- Jones, A., Buntings, C., & de Virus, M. J. (2013). The developing field of technology education: A review to look forward. *International Journal of Technology and Design Education*, 23(2), 191-212.
- Jones, A., & Moreland, J. (2004). Enhancing practicing primary school teachers' pedagogical content knowledge in technology. *International Journal of Technology and Design Education*, *14*(2), 121-140.
- Jones, E. (2013). First year teachers' technology use: Perceptions of factors affecting technology integration. Doctoral dissertation, University of Kansas.
- Kanjornboon, A. B. (2004). Creating useful knowledge: A case study of policy development in e-learning at Chulalongkorn University Language Institute. Melbourne, Australia.
- Keating, T., & Evans, E. (2001). Three computers in the back of the classroom: Pre-service teachers' conceptions of technology integration. Paper presented at the annual meeting of the American Educational Research Association, Seattle, WA.
- Khoza, S.B. (2001). The Outcomes of students studying a computer literacy course at Unischool. Masters dissertation, University of Durban Westville.
- Khoza, S. B. (2013b). Learning outcomes as understood by 'publishing research' facilitators at a South African university. *Mevlana International Journal of Education, 3*(2), 1-11.

- Kraak, A., & Hall, G. (1999). Transforming Further Education and Training (FET) in South Africa: a case study of Technical Colleges in Kwazulu-Natal. Pretoria: Human Sciences Research Council (HSRC) Press.
- King, K. (2011). Towards a new global world of Skills Development? TVET's Turn to make its mark. *NORRAG News, 46*(2), 1-11.
- Kvale, S. (1996). Interviews: An introduction to qualitative research interviewing. London: SAGE.
- Lebow, D., & Wager. (1994). Authentic activity as a model for appropriate learning activity: Implications for emerging instructional technologies. *Canadian Journal of Educational Communication*, *23*(3), 231-244.

Lincoln, Y., & Guba, E. (1985). *Naturalistic Inquiry*. London: SAGE.

- Longhurst, R. (2003). Semi-structured interviews and focus groups. In N. Clifford, S. French, & G. Valentine (Eds.), *Key methods in Geography* (pp. 117-132). London: SAGE.
- Lundeberg, M. A., Bergland, M., Klyczek, K., & Hoffman, D. (2003). Using action research to Develop preserve teachers' beliefs, knowledge and confidence about technology. *Journal of Interactive Online Learning, 1*(4). Retrieved from http://ncolr.uidaho.com/jiol/archives/2003/spring/toc.asp.
- Makgato, M., & Mbanguta, Z. (2002). Towards an outcomes based teaching and learning module for teacher preparation programmes of Further Education and Training institutions engineering and technology lecturer in South Africa. *World* Trans. on *Engineering and Technology Education*, 1(2), 227-232.

- Makoe, M. (2012). Teaching Digital Natives: Identifying competencies for mobile learning facilitators in distance education. *South African Journal of Higher Education, 26*(1), 91-104.
- Maree, K., Creswell, J., Ebersohn, I., Eloff, R., Ferreira, N., Ivankova, N., Jansen, J., Nieuwenhuis, J., Pietersen, V., & Plano Clark, C. (2007) *First Steps in Research*. Pretoria: Van Schaik.
- Margerum-Leys, J., & Marx, R. (2002). Teacher knowledge of educational technology: A study of student teacher/mentor teacher pairs. *Journal of Educational Computing Research*, 26(4), 427–462.
- McGrath, S. (2012). Vocational Education and Training for Development: A Policy in Need of a Theory? *International Journal of Educational Development*, 32(5), 623–631.
- McMillan, J.H., & Schumacher, S. (2010). *Research in Education: Evidence-Based Inquiry.* (Eds.). Virginia: Pearson education.
- Merriam, S. B. (1998). Qualitative research and case study applications in education. San Fransisco: Jossey-Bass.
- Mishra, P., & Koehler, M. J. (2006). Technological Pedagogical Content Knowledge: A new framework for teacher knowledge. *Teachers College Record*, 108(6), 1017-1054.
- Mouton, J. (1996). Understanding social research. Pretoria: Van Schaik.
- Mouton, J. (2001). *How to succeed in your master's and doctoral studies: A South African guide and resource book.* Pretoria: Van Schaik.
- Moon, J. (2002). *The module and programme development handbook.* London: Kogan Page Limited.

- Ndjabili, A. F. (2008). Curricula under pressure: aspects form the Eastern and Southern Africa region. *33*(6), 61-95.
- Neiss, M. L. (2005). Preparing teachers to teach science and mathematics with technology: Developing a technology pedagogical content knowledge. Teaching and Teacher Education, 21(5), 509–523.
- Newby, P. (2010). Research method for education. Harlow: Pearson education.
- Ngubane–Mokiwa, S. A. (2013). Information and communication technology as a learning tool: experiences of students with blindness. Doctoral dissertation, University of South Africa.
- Nguban-Mokiwa, S. A. (2014). Critiquing UNISA's Open Access Policy Through the Harry Frankfurt's Bullshit Lens. *Pensee Journal, 76*(9), 333-345.
- Ntshaba, L. P. (2012). Study of technology education instructional practices in grade nine classrooms: A case study of three senior secondary schools in the King Williams Town district. Doctoral dissertation, University of Fort Hare.
- Olsen, M.E., Lodwick, D.G., & Dunlap, R.E. (1992). Viewing the world ecologically. Boulder, CO: Westview Press.
- Pandor, N. (2006). Presentation of the Further Education and Training (FET) Colleges Act No. 16 of 2006. National Council of Provinces, October 2006.
- Park, M., & Sung, Y. (2013). Teachers' perceptions of the recent curriculum reforms and their implementation: What can we learn from the case of Korean elementary teachers? *Asia Pacific Journal of Education, 33*(1), 15-33.

- Peruski, L., & Mishra, P. (2004). Webs of activity in online course design and teaching. ALT-J, *Research in Learning Technology*, *12*(1), 37–49.
- Piaget, J. (1977). The development of thought: Equilibration of cognitive structures. NY: Viking Press.
- Pillay, J. (2012). Keystone Life Orientation (LO) teachers: implications for educational, social, and cultural contexts. South African Journal of Education.32(2), 167-177.
- Pinar, W. F. (2004). What is curriculum theory? In. W.F. Pinar, N.J. Mahwah, L. Erlbaum, *Studies in curriculum theory*, (Eds). NY: Routledge
- Polzer, T. (2007). Adapting to changing legal frameworks: Mozambican refugees in South Africa. *International Journal of Refugee Law, 19*(2), 22-50.
- Pool, J., Reitsma, G., & Mentz, E. (2013). An evaluation of Technology teacher training in South Africa: Shortcomings and recommendations. *International Journal of Technology and Design Education*, 23(2), 455-472.
- Powell, L. (2012a). Reimagining the Purpose of VET—Expanding the Capability to Aspire in South African Further Education and Training Students. *International Journal of Educational Development, 32*(5), 643-653.
- Powell, L. (2012b). Reimagining the Purpose of VET—Expanding the Capability to Aspire in South African Further Education and Training Students. *International Journal of Educational Development, 32*(5), 643-653.

Pretorius, J. (2000). Sociopedagogics. Pretoria: van Schaik.

- Punch, K. F. (2013). Introduction to social research: Quantitative and qualitative approaches. Thousand Oaks, CA: SAGE.
- Rand Afrikaans University. (2002). *Guideline on academics' ethics*. Johannesburg: RAU.
- Reeves, T. C., Herrington, J., & Oliver, R. (2004). A development research agenda for online collaborative learning. *Educational Technology Research & Development, 4*(52), 53-65.
- Rauner, F. (2005). Work process knowledge and the development of vocational competence. Paper delivered at the Vocational Education and Training conference. Proceedings of the 9th Conference. Muscat, Oman.
- Reid, M. S. (2000). Toward effective technology education in New Zealand. Journal of Technology Education, 11(2), 33-48.
- Rubin, H. J., & Rubin, I. S. (2011). *Qualitative interviewing: The art of hearing data*. Thousand Oaks, CA: SAGE.
- Rule, P., & John, V. (2011). Your guide to case study research. Pretoria: van Schaik.
- Rodríguez, A. G., & McKay, S. (2010). Professional Development for Experienced Teachers Working with Adult English Language Learners. CAELA Network Brief. Salt Lake City: Centre for Adult English Language Acquisition.
- Saunders, N.K. (2000). *Research Methods for Business Students (2nd Ed)*. Harlow, England: Financial Times/Prentice Hall.
- Schiro, M.S. (2013). *Curriculum theory: Conflicting visions and enduring concerns* (2nd ed.). Thousand Oaks, CA: SAGE.

- Schumacker, S., & McMillan, J. H. (1993). *Research in educaton. A conceptual introduction* (3rd ed.). New York: HarperCollins College Publishers.
- Shenton, A. K. (2004). Strategies for ensuring trustworthiness in qualitative research projects. *Education for Information*, 22(2), 63-75.
- Shulman, L. S. (1986). Those who understand: Knowledge growth in teaching. *Educational Researcher*, *15*(2), 4–14.
- Shulman, L. (2005). Signature pedagogies in the professions. *Daedalus*, 134(3), 52-59.
- Siskin, L. S. (2001). Subject divisions. Sociology of Education: Major Themes, *3*(61), 1420-1441.
- SLO (2008). Survey on the use of educational materials 2008/2009.Enschede: SLO.
- Smith, P., & Blake, D. (2005). Facilitating learning through effective teaching. At a glance. Retrieved from <u>http://www.ncver.edu.au</u>>.
- Stetsenko, A. (2008). From relational ontology to transformative activist stance on development and learning: expanding Vygotsky. *Cultural Studies of Science Education*, 2(3), 471-491.
- Strauss, A., & Corbin, J. (1998). Basics of qualitative research techniques and procedures for developing grounded theory (2nd ed.). London: SAGE.
- Struwig, M., Struwig, F., & Stead, G. (2001). *Planning, Reporting & Designing Research*. Cape Town: Pearson South Africa.
- Sutton, M., & Levinson, B. (2001). *Policy as practice: towards a comparative socio-cultural analysis of educational policy.* Westport: Ablex.

- Thomas, G. (2013). *How to do your research project: A guide for students in education and applied social sciences.* London: SAGE.
- Vygotsky, L. S. (1978). *Tool and Symbol in Child Development.* Cambridge: Harvard University Press.
- Wedekind, V. (2016). Literature on Technical and Vocational Education (TVET) College lecturers': a review. SAQA Bulletin. Waterkloof: Pretoria.
- Wedekind, V., & Watson, A. (2016). Understanding complexity in the TVET College system: an analysis of the demographics, qualifications, and experiences of lecturers' in sixteen TVET Colleges in Gauteng and KwaZulu Natal. SAQA Bulletin. Waterkloof: Pretoria.

Wedekind, V., & Watson, A. (2016). SAQA Bulletin. Waterkloof: Pretoria.

- Weimer, M. (2007). Intriguing connections but not with the past. *International Journal for Academic Development*, *12*(1), 5-8.
- Wilig, C. (2001). Introducing qualitative research in psychology: Adventures in theory and method. Great Britain: Open University Press.

Yin, C. (2014). Case study research: Design and methods. London: SAGE.

- Zhao, Y. (Ed.). (2003). What teachers should know about technology: Perspectives and practices. Greenwich, CT: Information Age.
- Zirkle, C. (2002). Using the Internet to enhance teacher education. *Techniques*, 77(5), 24-25.

APPENDIX A ETHICAL CLEARANCE



11 August 2016

Mrs Amanda Paul nee' Niemack (875874211) School of Education Edgewood Campus

Dear Mrs Paul nee' Niemack,

Protocol reference number: HSS/1170/016M Project title: Exploring lecturers' experiences of teaching Level 2 Life Orientation (Computer Skills) in National Certificate Vocational at a Technical Vocational Education and Training College

Full Approval – Expedited Application In response to your application received on 29 July 2016, the Humanities & Social Sciences Research Ethics Committee has considered the abovementioned application and the protocol have been granted FULL APPROVAL.

Any alteration/s to the approved research protocol i.e. Questionnaire/Interview Schedule, Informed Consent Form, Title of the Project, Location of the Study, Research Approach and Methods must be reviewed and approved through the amendment/modification prior to its implementation. In case you have further queries, please quote the above reference number.

PLEASE NOTE: Research data should be securely stored in the discipline/department for a period of 5 years.

The ethical clearance certificate is only valid for a period of 3 years from the date of issue. Thereafter Recertification must be applied for on an annual basis.

I take this opportunity of wishing you everything of the best with your study.

Yours faithfully

Dr Shenuka Singh (Chair)

/ms

Cc Supervisor: Mrs N Nzimande Cc Academic Leader Research: Dr SB Khoza Cc School Administrator: Ms Tyzer Khumalo

Humanities & Social Sciences Research Ethics Committee							
Dr Shenuka Singh (Chair)							
Westville Campus, Govan Mbeki Building							
Postal Address: Private Bag X54001, Durban 4000							
receptione: +27 (0) 31 260 3567/#39#4657 Pecamite: +27 (0) 31 260 4609 Email: https://www.ac.ze anymanm@ukzn.ac.ze anymanm@ukzn.ac.ze/ mphurp@ukzn.ac.ze/							
Website: www.ukan.ac.za							
1910 - 2010							
Frankting Congresses 🗰 Edgewood 💼 Howard College 😐 Medical School 🗰 Pietermaritzburg 💼 Westville							

APPENDIX B

INFORMED CONSENT OF RESEARCH PARTICIPANTS

Dear Sir/Madam

I, Amanda Paul, am currently a registered Masters student at the University of KwaZulu-Natal. As part of fulfilling my degree requirements, I am required to conduct a research project in my field of interest. The following topic is the focus of my research: Exploring lecturers" experiences of teaching Level 2 Life Orientation (Computer Skills) in National Certificate Vocational at a Technical Vocational Education and Training College. Through the use of lecturers, this study aims to answer the following critical questions:

1. What are lecturers' experiences of teaching Level 2 Life Orientation (Computer Skills) in the NVC curriculum?

2. Why do lecturers' have these experiences of teaching Level 2 Life Orientation (Computer Skills) in the NCV curriculum?

3. How do lecturers' experiences of teaching Level 2 Life Orientation (Computer Skills) in NCV impact their teaching?

Participation in this study requires you to

• Be interviewed on a one-on-basis

With your permission, both these interviews will be audio-taped.

As a participant

- Your involvement in the study is voluntary
- Your confidentiality will be maintained (all participants will use pseudonyms)
- You will not be exposed to any risk and you may choose not to answer questions should wish to do so
- You may withdraw at any time during the research process
- You will have access to your data with the rights to review, retract, revise your opinions
- You will have an opportunity to verify transcript/s of the one-on-one interview in order to confirm that it is your opinion and has been captured accurately
- You will receive a copy of all research reports and publications should you wish to

Permission to conduct this research study has been obtained from the University of Kwazulu-Natal. Should you have any questions about your participation and your rights in the study you may contact Ms Phume Ximba of UKZN Humanities and Social Sciences Research Ethics Committee at <u>ximbap@ukzn.ac.za</u> or call her at 27 31 2603587.

Thank you for your co-operation.

Amanda Paul

Contact Details:

Researcher: Amanda Paul

Supervisor: Nomkhosi Nzimande

Email address: <u>amandapaul269@yahoo.com</u> 0828853199 Nzimandem2@ukzn.ac.za 031-2603357

Informed consent of participant

I have read the information sheet and understand my participation in the study.

I understand that my real name will be not be used in all write-ups of this study and that the information that I will provide will be used for this research project and other appropriate research presentations. I am aware that:

- Participation is voluntary
- The interviews will be audio-taped
- I am not forced to answer any questions that make me uncomfortable and
- I am free to withdraw from the project at any point
- There is no payment for participation

I hereby consent / do not consent to have this interview recorded.

I hereby give consent to participate in this research project.

Signature:	Date:	

PRINCIPALS CONSENT LETTER APPENDIX C

Informed permission for the research from authorities

19 Fowey Avenue Bluff DURBAN

4052

The Principal

4000

Dear Sir/Madam

am Amanda Paul, currently a registered Masters student at the University of KwaZulu-Natal (Edgewood campus) in South Africa. A part of my development, I am undertaking a research study entitled 'Exploring lecturers' experiences of teaching Level 2 Life Orientation (Computer Skills) in National Certificate Vocational at a Technical Vocational Education and Training College'. I am seeking permission to interview five teachers from your College. Interviews will be conducted after College hours or during vacation at a venue convenient for the lecturers'. Therefore, the college will not be affected in anyway and the lecturer' will not be detracted from toing their duties at college.

The study does not seek any information about the College or about specific individuals, i.e. either colleagues or students. Its focus is on the lecturers' experiences at College. The study is mportant as it will reveal why lecturer' have these experiences and how these experiences nfluence their teaching.

³ermission to conduct this research study has been obtained from the University of KwaZuluvatal. Should you have any questions about its legitimacy, you can contact Ms Phume Ximba of JKZN Humanities and Social Sciences Research Ethics Committee (HSSREC) at umba@ukzn.ac.za or call her at +27(0) 31 2603587.

should you need further explanations or clarifications about the study, feel free to contact me or ny supervisor, Ms Nzimande. Our contact details are provided below.

our understanding and co-operation in this regard will be highly appreciated.

manda Paul itudent number: 875874211 lesearcher: Amanda Paul mail address: amandapaul269@yahoo.com Nzimandem2@ukzn.ac.za ell: 0828853199

Supervisor: MsNomkhosiVzimande 031-260 3357

Informed permission from Principal I have read and understood all the terms stipulated for the conduction of this study. I do/do not grant the researcher permission to conduct the study using lecturer' from this College Name: M Kug 2016 Date Signature: College stamp DEPARTMENT OF HIGHER EDUCATION & TRAINING 2 0 JUL 2018

APPENDIX D <u>SEMI-STRUCTURED INTERVIEW SCHEDULE</u>

I am Amanda Paul, currently a registered Masters student at the University of KwaZulu-Natal (Edgewood campus) in South Africa. As part of my development, I am undertaking a research study entitled 'Exploring lecturers' experiences of teaching Level 2 Life Orientation (Computer Skills) in National Certificate Vocational at a Technical Vocational Education and Training College'. The following questions will serve as a guide to our conversation.

Background information

When were you appointed to this College? Apart from Computer Skills, which other subjects are you teaching at present?

In which levels?

How long have you been teaching Computer Skills?

1. What are your experiences in teaching LO Computer Skills? (Resources available i.e. hardware and software)

2. What are your feelings towards training provided for Computer Skills lecturers' in order to achieve their goals? (Goals i.e. aims, objectives and outcomes)

3. What is your opinion about the benefits of training for Computer Skills lecturers'?

4. How would you describe your role as a lecturer teaching Computer Skills to students? (rationale-teaching reasons)

5. How would you rate lecturers' efforts in teaching Computer Skills? Explain your rating?

6. Would you say that lecturers have a responsibility in taking measures to improve their Computer Skills training? Why?

7. What do you think are the reasons for lecturers to have these teaching challenging experiences in Computer Skills? (If any)

8. In your opinion, does Department of Higher Education and Training provide any support for the training of lecturers in the teaching of the NCV in general or Computer Skills specifically?

9. Do experienced Computer Skills lecturers assist their fellow lecturers' in the teaching of Computer Skills? (With regard to content)

10. How does DHET monitor lecturers' performance in teaching of Computer Skills – component of LO? (Do they use the assessment marks as a benchmark?) (Are you aware of the purpose of assessments for students i.e. formative and summative?)

11. What support do you give to your Computer Skills Colleagues in order to assist them?

12. Are there support structures that have been implemented by the College in assisting lecturers with regard to Computer Skills? What kind of structures?

13. How would you rate your content knowledge used in Computer Skills? (Ms Word, Ms Excel, PowerPoint, Ms Access) Explain your rating?

14. Regarding the NCV LO Computer Skills curriculum...Is it teacher-centred, content-centred or learner-centred? Explain?

15. Do you know that the NCV LO Computer Skills curriculum demands a learner-centred approach to teaching? Would you say that in your teaching you follow these demands? To what extent?

APPENDIX E TURNITIN CERTIFICATE

) feedback studi	o Amanda Niemack Dissertation	/0		<	> (?
				Match Overvie	w	×
			8%			
			<			>
	3		1	uir.unisa.ac.za Internet Source	1%	>
	implementation of the curriculum as planned. Teaching resources like the		2	www.saqa.org.za Internet Source	1%	>
	availability of frained or qualified educators are just some of the issues highlighted by Ndjabili (2008). In the teaching of the Computer Skills component of Life	8	3	researchspace.ukzn.ac Internet Source	1%	>
	Orientation, a trained lecturer is essential.	T	4	scholar.sun.ac.za Internet Source	1%	>
	The NCV curriculum includes three mandatory subjects which are similar to the	0	5	www.jl4d.org Internet Source	1%	>
	school curriculum, namely; Mathematical Literacy or Mathematics, English and life Skills (life Orientation). These fundamental courses are core to all	<u>•</u> (<i>i</i>)	6	alternation.ukzn.ac.za Internet Source	1%	>
	specialisations and are "generic rather than specialised" (Wedekind & Watson, 2016, p. 78) An important issue in relation to lecturer identity is highlighted, as		7	Meahabo Dinah Magan Publication	<1%	>
¹ age: 8 of 89 Word	2016, p. 78) An important issue in relation to lecturer identity is highlighted, as and account: 22597	Text-only Report High Reso		On Q	< %	>